

Topics: Dual-purpose power plants District heating Cogeneration Design EPRI EM-3867 Volume 4 Project 1276-5 Final Report February 1985

# Design and Analysis of District Heating Systems

Volume 4: Assessment of District Heating for Springfield, Massachusetts

Prepared by Burns and Roe, Inc. Oradell, New Jersey

RN 6298 (3867,4)

	REP	ORT	S	UM	Μ	Α	R	Y	
SUBJECTS	Building energy	v systems / Cogener	ation/wa	aste heat re	ecover	у			
TOPICS	Dual-purpose p	ower plants	Cog	eneration					
	District heating		Des	ign					
AUDIENCE	Customer service managers / Generation planners								

## Design and Analysis of District Heating Systems

Volumes 1-4

Rising energy costs and dwindling fuel sources have spurred development of alternative energy-use technologies. One such technology, which reduces waste and increases the use of available energy supplies, is cogeneration of electricity and heat energy. This study evaluates utility cogeneration systems that cost-effectively provide district heating as well as electricity.

BACKGROUND

D District heating—the centralized production of heat energy and its distribution through pipes to provide space and hot water heating to homes, businesses, and industries within a certain area—is widely used throughout Europe, where dense population and relatively expensive fuel make it a viable technology. The need to conserve oil and gas and the success of European systems, which use hot water to distribute the heat, have sparked U.S. interest in district heating. However, research is necessary to assess the technical and economic feasibility of U.S. applications and to evaluate sitespecific designs.

OBJECTIVES • To assess the application of hot water district heating in the United States.

• To design district hot water systems for three utility sites and to evaluate the technical, economic, and environmental effects of these systems.

APPROACH To identify potentially attractive sites for cogeneration hot water district heating systems, project personnel developed a methodology for assessing potential heat sources and heat loads, analyzing hot water transmission and distribution pipe systems, determining the mode of operation of the system, identifying environmental and institutional issues, and analyzing economic factors. Using this methodology, they studied designs for power plant modifications and transmission/distribution piping systems for district heating in three candidate cities—Lansing, Michigan; Providence, Rhode Island; and Springfield, Massachusetts.

RESULTS

The study shows that an economically attractive site for district heating should have a high heat load density, usually between 60 and 90 MW/mi<sup>2</sup>; a short distance between the heat source and heat load; a coal-fired or **TIB/UB Hannover** 

FOC

89

30 644 730

refuse-derived-fuel-fired electrical generating station that could be modified to provide district heat along with electricity; and utility interest in owning and operating the system. Researchers produced site-specific conceptual designs for power plant modifications and transmission/distribution piping systems for the three candidate cities. They also devised plans for implementing the district heating systems and installing them in phases corresponding to increases in district load. In general, heat sources for the systems would be coal-fired or refuse-derived fuel-fired. The environmental assessment produced encouraging results: for example, replacement of numerous uncontrolled heat sources by a central cogeneration plant would improve local air quality. The economic analysis indicated that district heating would supply heat for downtown areas at a lower cost than individual boilers fired with oil or gas.

#### EPRI PERSPECTIVE

This study is part of EPRI's ongoing research in dual energy-use systems and their application in the United States. A related EPRI report, EM-2864, evaluates European district heating systems. In cooperation with Northeast Utilities, the city of Springfield, Massachusetts, is using the results of this research to carry out a more detailed analysis of the feasibility of district heating. The Lansing Board of Water and Light is using the project results to plan a district heating system for the downtown area.

Volume 1 of this four-volume report is an executive summary; Volumes 2, 3, and 4 are assessments of district heating for Lansing, Michigan; Providence, Rhode Island; and Springfield, Massachusetts, respectively.

#### PROJECT RP1276-5

EPRI Project Manager: S. David Hu Energy Management and Utilization Division Contractor: Burns and Roe, Inc.

For further information on EPRI research programs, call EPRI Technical Information Specialists (415) 855-2411.

ORDERING INFORMATION	EPRI EM-3867, Vols. 1–4, Final Report, February 1985. V1, 48 pages. V2, 92 pages. V3, 128 pages. V4, 104 pages.
EPRI Members	If this report is not available from your company libraries or you Technical Information Coordinator, you can order it from
	Research Reports Center P.O. Box 50490 Palo Alto, CA 94303 (415) 965-4081
Nonmember	S You can order this report in print or microfiche from Research Reports Center.
	Price: V1 \$8.50; V2 \$11.50; V3 \$14.50; V4 \$13.00 Overseas price: V1 \$17.00; V2 \$23.00; V3 \$29.00; V4 \$26.00 (California residents add sales tax.) Payment must accompany order.

## Design and Analysis of District Heating Systems Volume 4: Assessment of District Heating for Springfield, Massachusetts

#### EM-3867, Volume 4 Research Project 1276-5

Final Report, February 1985

#### Prepared by

BURNS AND ROE, INC. 800 Kinderkamack Road Oradell, New Jersey 07649

Principal Investigators I. Oliker F. Silaghy

UNIVERSITÄTSBIBLIOTHEK HANNOVER TECHNISCHE INFORMATIONSBIBLIOTHEK

Prepared for

Electric Power Research Institute 3412 Hillview Avenue Palo Alto, California 94304

> EPRI Project Manager S. D. Hu

Industrial Program Energy Management and Utilization Division

RN 6298 (3867,4)

#### ORDERING INFORMATION

Requests for copies of this report should be directed to Research Reports Center (RRC), Box 50490, Palo Alto, CA 94303, (415) 965-4081. There is no charge for reports requested by EPRI member utilities and affiliates, U.S. utility associations, U.S. government agencies (federal, state, and local), media, and foreign organizations with which EPRI has an information exchange agreement. On request, RRC will send a catalog of EPRI reports.

Copyright © 1985 Electric Power Research Institute, Inc. All rights reserved.

#### NOTICE

This report was prepared by the organization(s) named below as an account of work sponsored by the Electric Power Research Institute, Inc. (EPRI). Neither EPRI, members of EPRI, the organization(s) named below, nor any person acting on behalf of any of them: (a) makes any warranty, express or implied, with respect to the use of any information, apparatus, method, or process disclosed in this report or that such use may not infringe privately owned rights; or (b) assumes any liabilities with respect to the use of, or for damages resulting from the use of, any information, apparatus, method, or process disclosed in this report.

Prepared by Burns and Roe, Inc. Oradell, New Jersey

#### ABSTRACT

This report assesses the technical and economic feasibility of implementing a hot water district heating system in Springfield, Massachusetts. It provides conceptual designs for power plant modifications and transmission/distribution systems to satisfy a projected heat load. It includes a system implementation plan with equipment installation phases corresponding to projected increases in heat load. It also calculates the financial implications of such a hot water district heating system. This report is volume 4 of a 4 volume research project entitled "Design and Analysis of District Heating Systems." Volume 1 summarizes the methodology for assessment of district heating technical and economic feasibility and the results of studies performed for the cities of Lansing, Michigan; Providence, Rhode Island; and Springfield, Massachusetts. Volume 2 presents the site specific results for Lansing, Michigan and volume 3 presents the site specific results for Providence, Rhode Island.

#### ACKNOWLEDGMENTS

This study was performed in close cooperation with Northeast Utilities and the City of Springfield. Messrs. F. B. Hill, R. H. Meyer, F. P. Sabatino, J. J. Shea and C. F. Carlin of the former, and Messrs. J. J. Superneau and A. W. Curto of the latter were particularly helpful. Institutional aspects were investigated by Mr. E. J. Doyle, Jr. of Doyle Associates, Inc. CONTENTS

Section		Page
1	INTRODUCTION	1-1
2	HEAT LOAD ASSESSMENT	2-1
	Introduction	2-1
	Definition of Service Area	2-1
	Methodology for Determining Heat Load	2-1
	Result of the Heat Load Assessment	2-4
3	HEAT SOURCE ANALYSIS	3-1
	Introduction	3-1
	Natural-Gas-Fired High Temperature Hot Water Boilers	3-1
	Solid Waste Energy Recovery Facility	3-2
	Modification of Turbines at West Springfield Generating Station	3-3
4	TRANSMISSION AND DISTRIBUTION SYSTEM	4-1
	General	4-1
	Piping System Design	4-1
	Pipe Routing	4-5
	Transmission and Distribution System Costs	4-8
5	SYSTEM IMPLEMENTATION	5-1
6	SYSTEM OPERATION AND DISPATCH	6-1
	System Operation	6-1
	Plant Dispatch	6-6
7	ENVIRONMENTAL ASSESSMENT	7-1
8	INSTITUTIONAL ASSESSMENT	8-1
	Ownership Options and Financial Considerations	8-1
	Regulatory and Legal Aspects	8-10
9	ECONOMIC ANALYSIS	9-1

#### ILLUSTRATIONS

Figure		Page
2-1	Outdoor Temperature Profile	2-2
2-2	Prospective Service Area	2-3
2-3	Block Locations	2-6
3-1	SWERF Turbine Performance	3-4
3-2	Unit 1 and 2 Original Heat Balance	3-5
3-3	Unit 1 and 2 Modified Heat Cycle	3-7
3-4	Modified Unit 1 and 2 Turbine Performance	3-10
4-1	District Heating Conduit	4-4
4-2	Conduit Installation	4-4
4-3	Schematic of Transmission System	4-6
6-1	Heat Load Duration Curve for 1985 and 1986	6-2
6-2	Heat Load Duration Curve for 1987	6-2
6-3	Heat Load Duration Curve for 1988	6-3
6-4	Heat Load Duration Curve for 1989	6-3
6-5	Heat Load Duration Curve for 1990	6-4
6-6	Heat Load Duration Curve for 1991	6-4
6-7	Heat Load Duration Curve for 1992	6-5
6-8	Heat Load Duration Curve for 1993	6-5
9-1	Unit Cost of District Heat	9-4
9-2	Projected Heating Cost Comparison	9-4

## TABLES

Table		Page
2-1	Block Heat Loads	2-5
3-1	Unit 1 and 2 Turbine Cycle Modifications for District Heating	3-8
3-2	Plant Retrofit Costs	3-8
4-1	Transmission Piping System Dimensions	4-7
5-1	Development Strategy	5-2
5-2	Capacity Addition and Heat Load Development	5-3
5-3	Capital Cost Summary	5-3
6-1	System Annual Performance	6-7
6-2	WSGS Unit 1 Estimated Electrical Capacity Reduction	6-10
6-3	WSGS Unit 2 Estimated Electrical Capacity Reduction	6-10
7-1	Projected Displaced Fuel	7-2
9-1	Annual Carrying Charges, Phase One	9-5
9-2	Annual Carrying Charges, Phase Two	9-6
9-3	Annual Carrying Charges, Phase Three A (Municipal)	9-7
9-4	Annual Carrying Charges, Phase Three B (Utility)	9-8
9-5	Annual Carrying Charges, Phase Four	9-9
9-6	Annual Carrying Charges, Phase Five	9-10
9-7	Annual Carrying Charges, Phase Six A (Municipal)	9-11
9-8	Annual Carrying Charges, Phase Six B (Utility)	9-12
9-9	Annual Carrying Charges, Phase Seven	9-13
9-10	Annual Carrying Charges, Phase Eight	9-14
9-11	Composite Carrying Charges, Phases One Through Eight	9-15
9-12	Calculated Unit Cost for Hot Water District Heat	9-16
9-13	Hot Water Radiation Consumer Retrofit Payback	9-18
9-14	Two Pipe Steam Radiation Consumer Retrofit Payback	9-19
9-15	One Pipe Steam Radiation Consumer Retrofit Payback	9-20
9-16	Forced Air Consumer Retrofit Payback	9-21
9-17	Steam Air Handler Consumer Retrofit Payback	9-22

#### SUMMARY

#### BACKGROUND

This report is submitted under EPRI project RP-1276-5, Evaluation of Dual Energy Use Systems (DEUS) - Design and Analysis of District Heating Systems. It assesses the feasibility of hot water district heating in Springfield, Massachusetts, and develops such a system in conceptual form. The system considers the commercial, industrial and high density residential sections of the city as the heat load and a combination of high temperature hot water boilers, the Solid Waste Energy Recovery Facility and the West Springfield Generating Station as the heat sources.

The study assesses the available heat load, determines the available heat from the heat sources, develops a distribution network, develops a system implementation plan, assesses the environmental impacts, addresses institutional issues, and analyses the economics of the conceptual system.

#### HEAT LOAD ASSESSMENT

The heat load assessment determined the most suitable areas within the city for inclusion in the hot water district heating system and defined the heat load associated with these areas. Springfield has a population of about 152,000. It has a cold climate, with approximately 6000 heating degree days. The heat load analysis indicated a peak thermal demand of 98 MWt: 19.2 MWt in the commercial district, 31.5 MWt in the industrial district, and 47.3 MWt in the high density residential district.

#### HEAT SOURCE ANALYSIS

The combination of three heat sources is proposed for the Springfield district heating system.

A total of five approximately 11.7 MWt high temperature hot water boilers would be installed. During the initial phases of district heating development two or three of these boilers would satisfy the heat load. During the later phases and in the fully developed system, they would satisfy the peak loads only, while the West Springfield Generating Station and the Solid Waste Energy Recovery Facility would carry the district heating base load.

The turbine at the proposed Solid Waste Energy Recovery Facility would be adapted for district heating to generate hot water in a heat exchanger using steam from its extraction connection. A peak heat load of approximately 17.2 MWt could be provided at a loss of 2.1 MWe.

Two 50 MW turbines at the West Springfield Generating Station would be modified to extract steam from selected stages for use in new district heat exchangers. Each unit would be capable of supplying 22.2 MWt, with a corresponding loss in electrical generation of about 4.4 MWe.

#### TRANSMISSION AND DISTRIBUTION ANALYSIS

The proposed hot water district heating system will have a two-pipe closed transmission/distribution system with peak load supply and return temperatures of 250°F and 140°F.

The conduit system would consist of a carbon steel carrier pipe, polyurethane insulation, polyethylene casing, and leak detection system. Supply and return lines would be in the same trench. Installation of the distribution system will be beneath sidewalk areas.

Transmission/distribution system development would be based on a plan developed from an analysis of the heat load characteristics and the heat source alternatives. The objective would be to spread capital expenditures out over the development period, allowing the system to generate revenues to offset investments.

#### SYSTEM IMPLEMENTATION

The proposed 9-year development of the Springfield district heating system is planned to occur in eight phases, resulting in a system with a peak load rating of 98.64 MWt after 9 years.

In the first year of development, three high temperature hot water boilers would be installed at the City Hall boiler plant after the existing low capacity steam boilers were removed. Two of these generators would satisfy a developed heat load of 23.42 MWt. This phase would require the installation of about 4.5 miles of piping. During the second year, there would be no new heat source or district heating piping addition. Revenues would be collected to finance subsequent development.

In the third year, the district heating system load would be increased by an additional 17.23 MWt which would allow the addition of capacity from the turbine operating at the Solid Waste Energy Recovery Facility. This turbine would be operated as the base loaded heat source, while the high temperature hot water boilers would provide the peak heat loads and serve as backup. This phase requires the installation of supply and return pipes across the Memorial Bridge, together with the additional piping required for the added heat load.

During the fourth year the Unit 1 turbine at the West Springfield Generating Station would be modified and included in the district heating system. The total developed load during this phase would be 50.37 MWt. This unit, together with the turbine at the Solid Waste Energy Recovery Facility, would provide the base load.

Two more high temperature hot water generators would be added during the fifth and sixth years of implementation, together with additional piping necessary to increase the connected heat load to 64.88 MWt.

With the addition of another 24.16 MWt of heat load, the Unit 2 turbine would be included in the district heating system in its seventh year of development, providing a total installed capacity of 120.2 MWt.

In the eight and ninth years of system development, the piping system would be extended to connect remaining loads of 5.06 MWt and 4.54 MWt. In the ninth year, the system would be complete with a maximum connected load of 98.64 MWt.

#### ENVIRONMENTAL EFFECT

District heating would increase the thermal efficiency of all the turbine cycles operating in the district heating mode. The increased efficiency would result in a reduction in heat rejection to the environment per pound of fuel burned.

The district heating system would also affect air quality. The district heating system would replace numerous uncontrolled point sources with a few more efficient central sources. The net effect of the changes would depend on the changes in fuel use at the West Springfield Generating Station due to the district heating system. In this study it was assumed that the West Springfield Generating Station would be converted to coal and operated as a base loaded plant regardless of district heating, and thus the district heating system would not cause an increase in emissions at the station. During the course of the district heating study, Northeast Utilities plans for converting the Station to coal firing changed, coal conversion is no longer being actively pursued. The emissions from the Solid Waste Energy Recovery Facility are not assessed against the district heating since this plant must be operated regardless of a district heating system. However, the plant is provided with an air pollution control device which ensures that emissions are in conformance with current federal and/or state regulations applicable to this type facility. Emission from the gas-fired high temperature hot water generators may actually increase in the first few years, but in the fully developed system net gas consumption will decrease due to district heating, with a corresponding decrease in emission from gas sources.

#### INSTITUTIONAL ASSESSMENT

The institutional assessment for the Springfield district heating system examined a variety of options for the ownership, construction, and operation of the Solid Waste Energy Recovery Facility and district heating system. Other factors that were assessed include: system financing options, proposed federal legislation, regulatory and legal aspects, siting laws, and rate allocation. The assessment determined that there are no insurmountable obstacles to district heating in Springfield. However, many questions remain to be answered.

#### ECONOMIC ANALYSIS

The economic analysis was performed from the viewpoint of using the required revenue approach to determine the necessary charges for district heat. System implementation phases and unit costs of district heat are summarized in Tables S-1 and S-2.

Results of the economic analysis demonstrate that a hot water district heating system will supply heat at lower cost than individual boilers fired with oil (Figure S-1).

An analysis indicated payback periods varying from 3 to 6 years for most con-

sumers for district heating in comparison with 7.5% escalation of oil.

#### Table S-1

#### HOT WATER DISTRICT HEATING SYSTEM IMPLEMENTATION STATISTICS

	Phase One	Phase Two	Phase Three	Phase Four	Phase Five	Phase Six	Phase Seven	Phase Eight
Proposed Date On Line	1985	1987	1988	1989	1990	1991	1992	1993
Rated Peak Load (MWt)	23.42	40.65	50.37	61.99	64.88	89.04	94.1	98.64
Cumulative Annual Heat Sales (MWh) Investment (\$1983 x 10 <sup>3</sup> )	76490	132760	164500	202460	211900.	290800	307330	322160
Heat Source	1408	957	2216	470	470	2216	0	0
Piping	4347	2791	1583	1061	636	1855	741	445
Total	5755	3748	3799	1531	1106	4071	741	445

#### Table S-2

UNIT COST OF DISTRICT HEAT (\$/MBtu)

1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
12.33	12.98	8.08	8.54	8.74	9.36	10.09	10.53	10.99	11.43
1995	1996	1997	1998	1999	2000	2001	2002	2003	2004
11.88	12.43	13.07	13.80	14.54	15.44	16.25	17.32	18.75	19.06



Figure S-1. Projected Heating Cost Comparison

# Section 1 INTRODUCTION

This study develops in conceptual form a district heating system for the City of Springfield, Massachusetts. The study was performed in close cooperation with Northeast Utilities, with active participation of city development agencies and prospective customers.

Springfield, with a population of 152,319 is situated on the bank of the Connecticut River. During the 1960's and 1970's, Springfield suffered from a dwindling industrial base, increasing unemployment, and a migration to the suburbs by middle-income residents. The effects were particularly noticeable in the downtown area. From 1972 to 1977, Springfield's central business district suffered a 15.7% loss in the number of retail establishments and a 13.6% decline in employment.

Springfield has embarked upon an ambitious revitalization effort designed to attract commercial and retail interests, as well as residents to populate the downtown neighborhoods. These efforts have been supported by sources such as the Community Development Block Grant program, Economic Development Administration, Urban Development Action Grant program, and the Commonwealth of Massachusetts.

Springfield's leaders recognize that the municipality must continue its revitalization efforts if the city's renaissance is to be successful and that implementation of a district heating project would be an asset. The district heating concept has the potential to provide the low cost energy necessary to enhance Springfield's efforts in housing rehabilitation, commercial stabilization and industrial expansion.

The major components of a district heating system are the energy sources, transmission and distribution system, and users. The major heat sources considered are high temperature hot water (HTHW) boilers, a prospective solid waste energy recovery facility (SWERF) and two units of the West Springfield Generating Station (WSGS). Section 2 presents an assessment of the heat load, based on consumption records for existing buildings and city block maps of the prospective service areas. Section 3 characterizes the available heat sources and provides the composition and capital cost estimate of the ultimate heat source configuration. Section 4 presents the conceptual hot water transmission and distribution networks and assesses their costs. Section 5 describes the system implementation plan, and Section 6 discusses system operation and plant dispatch. Section 7 assesses the environmental impact of the prospective district heating system, and Section 8 provides an institutional assessment.

Section 9 presents the results of an economic analysis of the hot water district heating system based on budgetary capital costs, and operation and maintenance cost estimates.

suburbs by middle-income residents. The effects were particular y noticeable in the downtown area. From 1972 to 1977, Springfield's central business district soffered a 15.7% loss in the number of retail establishments and a 13.5% decline in employment.

Coringfield has embacked upon an embitious revitalization effort designed to attract connercial and retain interests, as well as residents to populate the downtown neighberhoods: "These efforts have been supported by cources such as the Community Development Block Grant program, itomenic Development Administration, Urban Development Action Grant program, and the Commonwealth of Massachusetts.

Springfield's leaders recognize that the municipality must continue its revitalization efforts if the city's remaissance is to be successful and that implementation of a district heating predect would be an associar when district heating concept has the potential to provide the low cost energy measurery to enhance Springfield's efforts in housing rehabilitation, commercial stabilization and industrial expansion.

The major components of a district heating system are the energy sources, transmission and distribution system, and users. The major heat sources considered are high temperature hot water (HTHH) boilers, a prospective solid waste energy recovery facility (SWERF) and two units of the West Springfield Generating Station (MSGS).

## Section 2 HEAT LOAD ASSESSMENT

#### INTRODUCTION

Major factors in a heat load assessment are the area heat load density and the proximity of the load to the heat source. For the progressive development of a district heating system in Springfield, it is assumed that HTHW boilers are the initial heat source, followed by the SWERF and two units at the WSGS. An area of adequate density to serve as a district heating load exists.

#### DEFINITION OF THE SERVICE AREA

Springfield is in southwestern Massachusetts, on the east side of the Connecticut River. The climate is cold, with approximately 6000 heating degree days. The outdoor temperature profile is shown in Figure 2-1. The outdoor design temperature is 3°F.

The primary heating district, closest to the heat sources, is the downtown central business area of about 157 acres. The HTHW boilers will be located in the City Hall boiler plant. The SWERF will be southwest of this district, and the WSGS is to the northwest, both on the other side of the river. The most favorable expansions for the district heating system would be to the north and east to the industrial district, and to the south to the high density residential areas. The entire prospective service area (Figure 2-2) occupies about 1210 acres.

#### METHODOLOGY FOR DETERMINING HEAT LOAD

The peak heat load estimate is based on block maps for each of the 75 assessor's blocks within the service area and building information such as area, height, condition, heating system type, fuel consumption, building function and construction type. All buildings within the potential service area are included in the heat load assessment. They include residential, commercial, institutional, public, hospital, public housing and other structures but no large factories.





The peak heat load estimate is based on block maps for each of the 75 assessor's blocks within the service drea and building information such us area, height, condition, heating system type, fuel consumption, building function and construction type. All buildings within the potential service area are included in the peat lead assessment. They include residential, commercial, institutional; public, hospital, public bousing and other structures but no large fac-





Selected major users in the service area were interviewed to gather data, including actual fuel use records. Building condition reports were provided by the Engineering Division of the City Department of Public Works for 1775 structures in the service area. In addition, 340 questionnaires were mailed to owners in the downtown area.

Heat loads calculated for buildings of known fuel consumption, internal temperature characteristics, and function were extrapolated to all similar buildings within the service area. Calculation of heat loads assumed an energy conversion efficiency of 60%.

The analysis assumed that all buildings which are currently vacant would be occupied and require heat at the time they were included in the system. The peak thermal requirements were estimated for the North Centre industrial site which is undergoing development.

#### RESULT OF THE HEAT LOAD ASSESSMENT

The results of the preliminary analysis indicate a peak thermal demand of approximately 98 MWt in the prospective service area. Individual block loads are listed in Table 2-1, Figure 2-3 shows the block locations.

The largest heat loads are the U.S. Armory (block 422) and in the North Centre industrial section (blocks 139 and 215). Other large consumers are the Pearl Street neighborhood and Springfield Quadrangle (with the city library and museums).

#### Table 2-1

Block Number	Peak Load (MWt)	Block Number	Peak Load (MWt)	Block Number	Peak Load (MWt)	Block Number	Peak Load (MWt)
140	0.87	309	1.94	414	3.94	615	0.22
141	0.91	310	0.81	415	1.29	616	0.51
142	0.98	311	0.73	419	0.75	617	2.15
143	0.88	312	0.99	420	2.05	618	0.80
156	0.31	313	1.08	421	0.45	619	0.87
157	0.75	314	0.52	422	14.85	620	0.93
202	0.05	315	1.13	503	0.92	621	0.61
203	0.09	316	1.48	508	1.11	622	0.61
204	0.04	401	1.12	509	0.55	623	1.13
207	0.31	402	1.50	518	1.63	712	0.57
208	0.10	403	1.05	607	1.04	720	1.00
214	0.13	404	1.23	608	1.29	721	0.30
216	0.90	405	0.77	609	0.28	727	0.88
217	0.45	406	0.86	610	0.79	737	0.72
304	1.10	407	0.39	611	0.08	738	0.41
305	0.42	408	1.77	612	0.69	739	1.65
306	0.93	409	1.28	613	0.98	744	0.53
307	1.29	410	1.34	614	0.60	139)	10.00
308	0.94	411	0.42			215)	19.60

#### BLOCK HEAT LOADS (98.64 MWt Total)



Figure 2-3. Block Locations

## HEAT SOURCE ANALYSIS

Section 3

#### INTRODUCTION

This section develops in a conceptual form the heat sources necessary for providing district heat in the form of hot water for Springfield. Three potential energy sources were investigated in sufficient detail to choose the best combination to supply the required energy in the most economical and reliable manner:

Constructing natural gas fired high temperature hot water boilers. Modifying the turbine at the planned SWERF (before installation). Modifying the turbines at the WSGS.

## NATURAL-GAS-FIRED HIGH TEMPERATURE HOT WATER (HTHW) BOILERS

To minimize capital expenditures for providing hot water district heat during the development stage of the system, HTHW boilers can be installed. Subsequently the boilers can be used for peaking and backup for the main energy sources.

The HTHW boilers would be located at the City Hall boiler plant where there will be sufficient room for the three initial units if the existing three small steam boilers are removed. The coal bunker areas adjacent to the boiler plant could be adapted to house two more units.

HTHW boilers are available in modules which require minimal field construction time. The proposed units are of the forced recirculation type with the following characteristics:

Generator capacity40,000,000 Btu/hrHeating capacity11.72 MWtDesign pressure275 psigOutlet temperature250°FReturn temperature140°FWater flow rate364,000 lb/hrBoiler efficiency at full load78.1%

Steam (Egm. erchermolitie haidtstilded is a starment better ab. 600 Asia and ASON and feedbare freedbare have been h





6631000 in hour interfection for the state of the heaviest in the open body in the featron density and the second state and the second state and the second state of the second state and the second s





3-5

following alternatives were considered:

- Modify the feedwater cycle to incorporate one or more district heat exchangers. These heat exchangers may be existing feedwater heaters serving a dual purpose, or new exchangers. This approach typically results in minimal loss in electrical generation, lowest capital costs and relatively small heat extraction compared with other retrofit schemes. This type of modification is useful as a first phase in the development of a large system.
- •) Maximize the extraction of steam from selected turbine stages and use the steam in new district heat exchangers. Two heat exchangers are usually arranged in series to increase the efficiency of heat extraction. For this type of retrofit, the throttle steam flow to the turbine is kept constant and the exhaust flow to the condenser is reduced. This alternative will supply substantial heat but may require permanent modifications to the turbine internals.
- •) Extract steam from an external turbine crossover pipe for use in district heat exchangers.

Analyses indicate that the preferred turbine modification would be to maximize selected extraction steam flows and use the steam in new district heat exchangers. The district heat exchangers would be outside the feedwater cycle, but the condensate would be returned to the cycle to minimize thermal loss and the need for make-up.

Figure 3-3 is a heat balance for the Unit 1 and 2 turbines modified for district heating. This mode makes available a maximum 22.2 MWt. The corresponding electrical output is about 45.6 MWe, compared with 50 MWe when no steam is being extracted for district heating. This electrical drop of 4.4 MWe translates to 1 MWe for every 5 MWt.

To produce this heat, about 43,800 lb/hr of steam is extracted from the 3rd point feedwater heater extraction at the slightly superheated condition of 263°F and 31.9 psia. This steam enters heat exchanger DH 1 where it raises the district heating water temperature from the 140°F return temperature to 205°F. The second stage heater, DH 2, raises the water temperature from 205°F to the 250°F supply temperature by extraction of about 29,800 lb/hr of steam from the 2nd point feedwater heater extraction at the superheated condition of 488°F and 121.5 psia. To transport 22.2 MWt, a flow rate of about 684,500 lb/hr of water is necessary. Unit 1 and 2 turbine performance with and without steam extraction for district heating is shown in Table 3-1. Electrical output as a func-



3-7

#### Table 3-1

UNIT	1	AND	2	TURBINE	CYCLE	MODIFICATIONS	FOR	DISTRICT	HEATING

	Original	Modified
Throttle Steam Conditions	Lasta in a spinistic of	
Flow, lb/hr	438,200	438,200
Temp., °F	900	900
Pressure, psig	850	850
Generator Output, MWe	50	45.6
District Heat Output, MWt District Heat Water	0	22.2
Flow Rate, 1b/hr	a increase. The strict	684,500
Supply Temp., °F	ofit, the throttal Th	250
Heat Rejected in		
Condenser, Btu/hr	2.96x108	2.35x108

#### Table 3-2

#### PLANT RETROFIT COSTS (1983 dollars)

ltem	\$/Unit
Turbine-Generator District Heat Exchangers District Heat Pumps Piping and Valves Make-up and Water Treatment Electrical Items Direct Cost	\$1,080,000 63,000 122,000 344,000 127,000 <u>37,000</u> \$1,773,000
25% Engineering, Construction Management and Contingency	443,000
Total	\$2,216,000

Note: Owner costs are not included in the above estimate.

tion of heat output is shown in Figure 3-4. For Unit 3, the extractions serving the 3rd and 4th point feedwater heaters were maximized. (Tapping into the crossover would be too costly because the crossover is internal.) Although the electrical rating for Unit 3 is twice that of Units 1 and 2, the available district heat is slightly less because of the physical characteristics of the available turbine stage nozzles and pressures. Therefore, Unit 3 was not chosen for modification.

In all cases of increased steam extraction, less steam is available throughout the turbine, and therefore less heat is transferred to the feedwater in the low pressure heaters. This reduction in feedwater temperature results in increased extraction in the high pressure feedwater heaters. Nevertheless, the final feedwater temperature entering the boiler may be slightly reduced. The redistribution of extraction flows in the turbine stages necessitates replacement of selected diaphragms and machining of shell internals. Thermodynamic and mechanical constraints must be assessed by the turbine manufacturer. These constraints include the maximum heat that can be extracted, blade loading as a result of increased extraction, extraction line limitations, steam flow and pressure changes in turbine elements, windage losses in LP elements, and thrust considerations.

#### Equipment Selection

In addition to the Unit 1 and 2 turbine modifications, the modification of the WSGS to district heating supply would involve installation of district heating equipment, including two district heat exchangers, extraction and drain piping, drain pumps, valves and controls, three 50% capacity circulating water pumps and piping, an expansion tank, and a make-up water system. The make-up system components are sized to produce treated water at the rate equal to about 0.5% of the district heating water flow rate. There would be two 100% capacity soft water pumps, one soft water storage tank, one 100% capacity vacuum degasifier, two 100% capacity vacuum degasifier pumps, and two 100% capacity zeolite softeners.

#### Plant Retrofit Costs

Table 3-2 shows \$2,216,000 estimated installed cost for proposed district heating modification at the WSGS.





## TRANSMISSION AND DISTRIBUTION SYSTEM

Section 4

#### GENERAL

This section develops in conceptual form a transmission and distribution system to integrate the Springfield heat load with the available heat sources. Development of the transmission and distribution system would be performed in phases to coincide with development of the heat sources. The transmission and distribution system is designed as a two pipe closed system for district heating water circulated by horizontal split case pumps located at each heat source.

#### PIPING SYSTEM DESIGN

District heating piping sizes are determined by the heat load and the water temperature drop in the system, which depend on the type of heat supply control. Simplest and most widely used in a central temperature control system in which the required amount of heat is provided by varying the temperature in the supply network at the heat source as a function of the outdoor temperature.

A large temperature drop between supply and return lines minimizes the flow rate of circulating water, piping size and pumping power. On the other hand, a high temperature of the supply water requires higher steam extraction pressure, which results in reduced plant electrical generation. Therefore, the economics of both the power plant and the piping system must be considered. The piping system for the district heating system is designed for peak supply temperature of 250°F and return temperature of 140°F.

As discussed in the Assessment of European District Heating Technology (EPRI EM-2864, February 1983), there are several types of piping system designs available for district heating systems. Conduit design is considered best suited to the proposed system. The conduit system consists of a thin-walled carbon steel carrier pipe, polyurethane insulation, polyethylene casing, and a leak detection system.

The effects of accelerated corrosion of carbon steel piping in the 200°F range require that the supply piping be physically protected against outside water

contact. Equally important, the pipe insulation must be protected from moisture penetration that would increase the thermal conductivity of the insulation. The insulation must also be protected against mechanical abuse, ultraviolet degradation, erosion and other environmental hazards. The polyurethane foam insulation must be protected against mechanical damage from weathering, handling abuse, field bending, casing pressures, water pressure, saddle-bearing forces, ground movement and damage in storage. The most economic and reliable method for providing the piping system with protection against the above problems is to use a shop-fabricated conduit design with a polyethylene casing.

A shop-fabricated conduit system can be fully shop tested, is less expensive to install, is easier to install causing less disruption of the district heating site, and can be manufactured with a high degree of quality control resulting in a product of high reliability. Polyethylene is the preferred casing material because it will not corrode in the underground environment, has excellent waterresisting properties, is easy and cost effective to install, and is light weight and easy to repair if physically damaged.

Polyurethane foam is preferred because of its low thermal conductivity (0.18 Btu/in./hr/ft<sup>2</sup>/°F) which provides excellent thermal efficiency of the conduit system. During manufacture of the conduit, the foam is poured between the carrier pipe and the casing in a liquid form and expands to occupy the space between the carrier pipe and the casing. The foam bonds itself to the pipe and casing walls, providing a structurally stable insulation that will not shift, settle or shrink with time to create voids. The conduit requires no special hardware for fastening the insulation in place.

The prefabricated conduit is installed by butt-welding the carrier pipe sections together; the joint is then coated with an elastic sealing compound. A steel fitting coated with a thick layer of polyethylene is bolted around the joint. For large diameter pipe, an aluminum jacket is placed around the joint and riveted in place instead of a bolted steel fitting. Polyurethane is then poured through holes in the aluminum to insulate the joint. The holes in the aluminum jacket are plugged, and a shrink sleeve is placed over the aluminum and heated to fit tightly.

The conduit is manufactured and shipped to the field in lengths of 20 feet or more. All fittings and valves are also preinsulated and shipped in the form of

a finished conduit section. Figure 4-1 shows the conduit design.

The small customer branches are conduits of the same designs, with flexible copper carrier pipe delivered in coils. Such conduit has the advantages of being flexible and requiring fewer joints.

To protect the system and facilitate service, a combined alarm and fault locator system is built into the conduit during manufacture. The alarm system consists of two copper wires molded into the foam insulation during manufacture of the conduit. As the conduit sections are installed, the wires are connected, forming a continuous circuit through the piping system. The copper wires are connected to alarm/control boxes at either central or remote locations. The wires carry a low voltage current. When moisture enters the piping system, it completes the circuit between the two wires and triggers an alarm. The alarm system not only detects the leak but, by measuring resistance, locates the leak.

Two different schemes were investigated for the location of the conduit: below the streets and below the sidewalks. Installing the conduit below the sidewalks has the advantage of requiring shallow excavation, typically resulting in a less expensive system than street installations. Sidewalk installations require less excavation and backfill, cause minimum disruption of traffic in congested areas, and usually involve fewer interferences with other underground utilities. Examination of available utility maps for Springfield determined that a minimum number of interferences with other utilities would occur if the district heating pipes were installed beneath the sidewalks.

The proposed technique is to install the supply and return piping side-by-side in a single trench beneath the sidewalk. The trenches will usually be less than 5 feet deep and require no shoring. The conduit is laid directly in the trench on a 4-inch sand bed. The excavation provides for a clearance of 6 inches between conduits and 6 inches between the conduit and trench wall. The conduit is covered with at least 18 inches of homogeneous stone-free sand. The sidewalk bed and pavement are installed directly over this sand. Figure 4-2 shows the recommended conduit installation, conduit diameters and required trench sizes for different size carrier pipes.

To minimize the cost of expansion provisions, the steel piping system expansion will be accommodated by a European technique utilizing the mechanical plasticity





Figure 4-2. Conduit Installation
of the pipe at stress ranges up to the elastic limits. A steel pipe is anchored at one end, and heated from the installation temperature of 65°F to about 120°F. The pipe is then anchored at the opposite end; this does not change the physical state of the pipe, and the tension in the steel will be zero. When the temperature of the pipe is increased to the operating temperature of 250°F, a compressive stress will be generated in the steel pipe within allowable structural limits. When the temperature is reduced, a tensile stress within the allowable limits will be generated. Special pipe fittings allow the system to be installed in accordance with this procedure.

Expansion loops or pipe offsets are not recommended because of the narrow access route that is available. Mechanical expansion joints of the slip, bellows, or ball/swivel type are not recommended because of their maintenance and installation requirements.

### PIPE ROUTING

To determine the optimal routing for the district heating piping, utility maps showing the locations of the existing underground utilities were examined, together with maps showing the blocks for which heat loads were determined. The required piping diameters and lengths between the various nodes were than calculated.

Figure 4-3 shows the proposed transmission system throughout the district heating service area, and Table 4-1 lists the associated pipe sizes and lengths. The principal transmission pipes will be routed along Main and Chestnut Streets in the north-south direction, with others in Pearl, Liberty, and State Streets in the east-west direction. Between nodes 1' and 1 there will be a pair of nominal 16-inch pipes. The pipes on Memorial Bridge over the Connecticut River will be 12-inch diameter.

Hot water will pass from the transmission mains through a distribution network to the building systems. Although it has been assumed that distribution piping would be located under the sidewalks, some piping may be routed through the basements of adjacent buildings in order to minimize system cost.



Figure 4-3. Schematic of Transmission System

#### Table 4-1

.

Nodes	Length (ft)	Nominal Dia(in.)	Nodes	Length (ft)	Nominal Dia(in.)
SWERE_P	1500	8	6-7	250	0.75
D 1	1875	12	7-8	250	0.75
1' 1	1500	16	13-19	875	8
1 / 2	625	8	19-20	625	8
1-45	913	8	19_21	938	4
43-44	913	8	21_22	1250	3
44-40	250	8	22-23	1438	2
45-40	1125	12	16 17	1063	5
1-9	1125	12	40-47	313	5
9-32	625	0	4/-40	500	4
32-33	025	0	40-49	375	4
33-34	313	8	49-50	120	4
9-10	188	8	50-51	430	2 5
10-11	500	8	51-52	430	2.5
11-12	500	8	52-53	100	2
12-13	375	8	34-37	250	4
34-35	2375	6	37-38	313	4
35-36	563	2	38-39	438	4
9-24	750	5	39-40	188	3
24-25	250	5	40-41	1250	2.5
25-26	313	2.5	40-42	500	2
26-27	625	1.5	46-54	500	5
25-28	563	4	54-55	688	5
28-29	1060	4	55-56	813	5
29-30	563	2	56-57	938	5
30-31	2125	1.25	57-58	250	4
13-14	188	2	58-59	313	3
14-15	563	2	59-60	1000	2.5
15-16	1250	2.5	60-61	625	2.5
16-17	750	1.5	61-62	500	2
17-18	563	1.25	57-63	250	4
1_2	188	4	63-64	313	3
2_3	500	2.5	64-65	250	3
2-5	375	2	65-66	750	3
1 5	313	1	66-67	563	2.5
4-5	625	0.75	67-68	750	2.5

### TRANSMISSION PIPING SYSTEM DIMENSIONS

in 1967, the turbline at the SWERS would be connected to the district beating system to supply an additional 17.2 NML capability (52.36 MML total). This would require a capital expenditure for heating equipment as well as for ofpin ocross Remorial bridge. The district heating network would be extended by the addition of a piping bettion between nodes 34 and 36 on State Street. The load would be 40.65 MML, and the total annual district heat sales would be 132.762 MML.

#### TRANSMISSION AND DISTRIBUTION SYSTEM COSTS

Estimates of piping system costs were developed by individually estimating the cost of each component and installation step. The cost estimate includes:

- Removal of existing sidewalk surface .
- . Excavation of the trench
- Hauling of excavated and backfill material .
- Spreading and compaction of backfill material .
- Prefabricated conduit and field joints .
- Pipe installation and testing .
- Isolation valves, fittings and specialties .
- Contingency for problems with underground construction .

Costs for piping and other materials are based on budget estimates received from vendors. Costs for earthwork and labor are based on reference sources for the construction industry. The transmission and distribution piping costs for the entire district heating system are estimated at \$10,120,000 and \$3,339,000 respectively, a total of \$13,459,000.

## Section 5 SYSTEM IMPLEMENTATION

The development of the Springfield district heating system is projected to occur over a 9-year period that will allow capital expenditures to be spread out in incremental investments and the system to generate revenues to offset the required capital investments. This development period will also provide the required lead time for system engineering, design and construction.

The general procedure will be to develop the heat load at an average increment of about 11 MWt per year. When sufficient heat load is developed, heat sources will be brought on line. The development sequence and corresponding capital costs are summarized in Tables 5-1 through 5-3.

In 1985, transmission and distribution piping would be installed to provide heat for parts of the central business district and industrial district. The initial connected load would be about 23.42 MWt. This would require about 4.5 miles of pipe including piping from the City Hall boiler plant to the end of Pearl Street (node 31) and from the intersection of Pearl and Chestnut Streets (node 9) to nodes 18 and 34, as well as extensions from this main route to nodes 4 and 27. Three HTHW boilers would be installed in the City Hall boiler plant. Two boilers would operate to supply heat to the network described, the third would be a spare. The total installed capacity would be 35.16 MWt. Annual district heat sales for this phase are estimated at 76,490 MWht.

In 1986, there would be no additions to the district heating system. Instead, revenues will be collected to help offset the 1987 capital expenditures. Total heat sales would remain 76,490 MWht.

In 1987, the turbine at the SWERF would be connected to the district heating system to supply an additional 17.2 MWt capability (52.36 MWt total). this would require a capital expenditure for heating equipment as well as for piping across Memorial Bridge. The district heating network would be extended by the addition of a piping section between nodes 34 and 36 on State Street. The load would be 40.65 MWt, and the total annual district heat sales would be 132,762 MWht.

#### TRANSMISSION AND DISTRIBUTION SYSTEM COSTS

Estimates of piping system costs were developed by individually estimating the cost of each component and installation step. The cost estimate includes:

- Removal of existing sidewalk surface
- Excavation of the trench
- Hauling of excavated and backfill material
- Spreading and compaction of backfill material
- Prefabricated conduit and field joints
- Pipe installation and testing
- Isolation valves, fittings and specialties
- Contingency for problems with underground construction

Costs for piping and other materials are based on budget estimates received from vendors. Costs for earthwork and labor are based on reference sources for the construction industry. The transmission and distribution piping costs for the entire district heating system are estimated at \$10,120,000 and \$3,339,000 respectively, a total of \$13,459,000.

#### Section 5

### SYSTEM IMPLEMENTATION

The development of the Springfield district heating system is projected to occur over a 9-year period that will allow capital expenditures to be spread out in incremental investments and the system to generate revenues to offset the required capital investments. This development period will also provide the required lead time for system engineering, design and construction.

The general procedure will be to develop the heat load at an average increment of about 11 MWt per year. When sufficient heat load is developed, heat sources will be brought on line. The development sequence and corresponding capital costs are summarized in Tables 5-1 through 5-3.

In 1985, transmission and distribution piping would be installed to provide heat for parts of the central business district and industrial district. The initial connected load would be about 23.42 MWt. This would require about 4.5 miles of pipe including piping from the City Hall boiler plant to the end of Pearl Street (node 31) and from the intersection of Pearl and Chestnut Streets (node 9) to nodes 18 and 34, as well as extensions from this main route to nodes 4 and 27. Three HTHW boilers would be installed in the City Hall boiler plant. Two boilers would operate to supply heat to the network described, the third would be a spare. The total installed capacity would be 35.16 MWt. Annual district heat sales for this phase are estimated at 76,490 MWht.

In 1986, there would be no additions to the district heating system. Instead, revenues will be collected to help offset the 1987 capital expenditures. Total heat sales would remain 76,490 MWht.

In 1987, the turbine at the SWERF would be connected to the district heating system to supply an additional 17.2 MWt capability (52.36 MWt total). this would require a capital expenditure for heating equipment as well as for piping across Memorial Bridge. The district heating network would be extended by the addition of a piping section between nodes 34 and 36 on State Street. The load would be 40.65 MWt, and the total annual district heat sales would be 132,762 MWht.

### Table 5-1

### DEVELOPMENT STRATEGY

Construc-	Transmiss Constru	ion Sys. ction	Section 5				
tion Year	From	То	Blocks Served (	HTHW Unit Construction	SWERF WS	Con GS Load	(MWt)
1985	HTHW 1 9 1 9 9 25	1 9 31 4 18 34 27	156,157,207,214, 216,217,304,305 309,313,315,316 401,402,403,404 405,406,414,415 508,509	, Unit 1 , Unit 2 , Unit 3	ent or the rectod t investagets ttal inves t the to	devēlopn a S-yea emental ired cap tred jaa	23.42
1986	Costs for	ear Times	Same as above	ased of rate			23.42
1987	SWERF P 34	Р 1' 36	419,422,518	43.191.1/18 Genetic 1.21 Second 1.2	SWERF turbine	6422988 6465088 1016 56	40.65
1988	1 46	46 49	306,307,308,310 311,312,410,411 612,613,614	n Ta-les 5-1	i bes-noter	Unit 1	50.37
1989	34 40 50	42 41 53	314,407,408,409 420,421,615,616 617,618,619,621	, Unit 4	the centration	parts of	61.99
1990	46 5	56 8	202,203,204,208 607,608,609	, Unit 5	ng piping d from the	ibulani ns (10 e	64.88
1991	13 19	23 20	139,140,141,142 143,215,503	il as extend nd bo testal	34, <u>35 we</u> offers wou	Unit 2	89.04
1992	57	68	610,611,727,733 738,739,744	to supply h	d operate The total	trov and	94.1
1993	57	62	620,622,623,712 720,721,202,203	das are est	14 erns 10	1 29/57	98.64

In 1987, the turbine at the SWERF would be connected to the district heating system to supply an additional 17.2 Mm capability (52.36 MM total). this would require a capital expenditure for heating equipment as well as for pipto across Mmmorial Bridge. The district heating network would be extended by the addition of a piping section between nodes 34 and 35 on State Street. The load would be 40.65 MM , and the total annual district heat sales would be

# Table 5-2

		Capability (MW	;)	Load	(MWt)	
Year	Addition	Installed	Available	Developed	Connected	
1985	35.16	35.16	23.44	23.42	23.42	
1986	0	35.16	23.44	0	23.42	
1987	17.2	52.36	40.64	17.23	40.65	
1988	22.2	74.56	52.36	9.72	50.37	
1989	11.72	86.28	64.08	11.62	61.99	
1990	11.72	98.00	75.80	2.89	64.88	
1991	22.2	120.2	98.00	24.16	89.04	
1992	NON COST PURTS	120.2	98.64	5.06	94.1	
1993	Million California	120.2	98.64	4.54	98.64	

# CAPACITY ADDITION AND HEAT LOAD DEVELOPMENT

### Table 5-3

# CAPITAL COST SUMMARY Projected Investment (\$1983 x 10<sup>3</sup>)

Year of Expenditure	1985	1986	1987	1988	1989	1990	1991	1992	1993	Total
HTHW Station	1408	0	0	0	470	470	0	0	0	2,348
SWERF	0	0	957	0	0	0	0	0	0	957
Power Plant Retrofits	0	0	0	2,216	0	0	2,216	0	0	4,432
Transmission System	3,555	0	2,208	1,254	668	538	1,038	567	292	10,120
Distribution System	792	0	583	329	393	98	817	174	153	3,339
Total	5,755	0	3,748	3,799	1,531	1,106	4,071	741	445	21,196

In 1988, one turbine at the WSGS would be modified to supply 22.2 MWt of district heat (74.56 MWt total). The district heating piping system would be extended by a section on Main Street between nodes 1 and 46 with a branch between nodes 46 and 49. during this phase, the HTHW boilers would operate mostly in the peaking mode. The connected load during this phase would be 50.37 MWt, and the estimated heat sales would be 164,508 MWht. In 1989, another HTHW boiler would be added at City Hall (86.28 MWt total). Piping would be installed from nodes 34 to 41 and 42, and from 50 to 53. Total connected load would reach 61.99 MWt, and total heat sales would be 202,458 MWht.

In 1990, a fifth and last HTHW boiler would be installed at City Hall (98 MWt total). Two sections of district heating piping would be added on Main Street (between nodes 5 and 8 and between nodes 54 and 56), bringing the connected load to 64.88 MWt. Total heat sales would be 211,897 MWht.

In 1991, the second turbine at the WSGS would be modified, and the 22.2 MWt output would be added to the system for a total capacity of 120.2 MWt. At the same time, the Liberty Street piping extension (between nodes 13 and 23) would be added to increase the heat load by 24.16 MWt. The total connected load would be 89.04 MWt, and the total heat sales would be 290,804 MWht.

In 1992, a section of piping will be installed between nodes 57 and 68, with an increase in heat load of 5.06 MWt. Thus the maximum rated heat load would then be 94.1 MWt and the heat sales would be 307,330 MWht.

In 1993, the last piping section, between nodes 57 and 62, would be installed, with a heat load addition of 4.54 MWt. The system would be complete, with a maximum connected heat load of 98.64 MWt and total estimated heat sales of 322,157 MWht. In this phase and thereafter, the system would operate with the SWERF and WSGS turbines as the base loaded energy sources and with the HTHW boilers supplying backup and peak loads.

In 1988, one turbine at the VSGS would be modified to supply 22.2 MML of district heat (74.56 MML total). The district heating piping system would be extended by a section on Main Street between nodes I and 46 with a branch between nodes 46 and 49. during this phase, the MTMM boilars would operate mostly in the peaking mode. The connected load during this phase would be 50.37 MML, and the estimated heat sales would be 164.508 MML

### Section 6

### SYSTEM OPERATION AND PLANT DISPATCH

### SYSTEM OPERATION

During a year, the actual heat demanded by the connected load varies inversely with the outdoor temperature. In order to determine the annual heat generation required by a district heating system, a heat load duration curve is required.

Using an outdoor minimum design temperature of 3°F, an indoor design temperature of 65°F, and the annual temperature profile, the annual heat load duration curves were developed for each phase of implementation (Figure 6-1 through 6-8). Total annual heat consumption at the ultimate connected peak load of 98.64 MWt is calculated to be approximately 322,160 MWht.

The heat load duration affects operation of a cogenerating unit since the maximum extraction for district heating which occurs at 100% heat load reduces electrical generation to its minimum. As the heat load decreases, steam extraction also decreases, resulting in higher electrical generation.

In the completed district heating system, the mode of operation of the system depends on the system load demand. In summer, when the load is generally constant and equal to the domestic hot water load, the system would operate at supply and return temperatures of 160°F and 104°F. The summer load is about 7.3 MWt, with a duration of about 3160 hours.

Planned maintenance on the modified WSGS units and the SWERF equipment should occur in summer. The HTHW heaters need not be used in summer since maintenance of the WSGS turbines and the SWERF can be staggered. For the purposes of this study, a maintenance downtime of 1 month for the SWERF has been assumed. Summer load can be carried entirely by the SWERF turbine.

6-1



Figure 6-2. Heat Load Duration Curve for 1987

.











Figure 6-5. Heat Load Duration Curve for 1990



Figure 6-6. Heat Load Duration Curve for 1991







Figure 6-8. Heat Load Duration Curve for 1993

During spring and fall, up to about 39.4 MWt space heating and domestic hot water loads can be satisfied by the SWERF turbine and one WSGS unit. The SWERF would operate in the base loaded mode, and the WSGS unit could satisfy the balance of the load with minimum loss in electrical output.

At heat loads between about 39.4 and 61.6 MWt, operation of both WSGS units and the SWERF would be necessary. Operation of both WSGS units at their maximum district heating output would be necessary for about 1700 hours annually. Between fall and spring, the district heating system supply temperature varies from 160°F to 250°F, and the return water temperature varies between 104°F and 140°F.

When the heat load in winter increases above about 61.6 MWt, one or more HTHW boilers would be placed in operation, depending on the heat load. Both WSGS units and the SWERF would operate in the base loaded mode. The district heating water supply and return temperatures would be 250°F and 140°F. Temperature would be held constant, and the flow rate and the number of operating HTHW boilers would be varied to supply the required load.

Using the cumulative durations of the various heat loads, the lost electrical generating capacity was calculated based on the performance of the modified turbines. The auxiliary power for pumping was also considered. The annual performance of the district heating system during its development is summarized in Table 6-1.

### PLANT DISPATCH

Power plants in the Northeast, including Northeast Utilities plants, are grouped to form the New England Power Pool (NEPOOL). The operating facility of NEPOOL, commonly referred to as the New England Power Exchange (NEPEX), dispatches the six-state area on a one-system basis in the most reliable and economic manner possible. NEPEX records the actual operation of each unit on an hourly basis and computes an individual hourly dispatch for each participating utility. On this basis, a utility's generating entitlements are dispatched against its own load. This hypothetical, paper computation is known as the "own-load" dispatch. The cost difference between the sum of all the participants' hour-by-hour own-load dispatches and the actual dispatch (less NEPOOL expenses) determines the savings generated from this pooling arrangement (known as NEPEX Savings Fund). Table 6-1

### SYSTEM ANNUAL PERFORMANCE

Operating Year	1985	1986	1987	1988	1989	1990	1991	1992	1993
Peak Connected Load (MWt)	23.42	23.42	40.65	50.37	61.99	64.88	89.04	94.10	98.64
Annual Heat Consumed (MWht)	76490	76490	132762	164508	202458	211897	290804	307330	322157
HTHW Generator Heat Production (MWht)(1)	80315	80315	38694	4607	18418	23562	18866	24841	32322
SWERF Heat Production (MWht)(1)	0	0	100706	107139	111981	112850	117924	118895	119762
WSGS Heat Production (MWht)(1)	0	0	0	60987	82181	86080	168553	178960	186180
Annual Lost Electrical (MWh)	664	664	1152	13305	17176	18042	34149	36202	37924

(1) Includes 5% transmission line loss

(2) Includes auxiliary power required for pumping

Accordingly, the energy cost to a NEPOOL participant (and ultimately its customers) is primarily determined by the energy cost of its own-load dispatch, computed as if it were isolated from the pool. However, these costs are reduced somewhat by the participant's share of the NEPEX Savings Fund. Northeast Utilities planning is therefore based on the expected own-load energy costs, modified by an allowance for NEPEX savings.

The total system energy cost of the Northeast Utilities system companies can be very closely approximated by simulation of a dispatch in which the companies' wholly owned units and entitlements in other generating units are economically operated to supply the hour-by-hour energy requirements of its own system. This dispatch of the Northeast Utilities system companies is simulated by using a computer model called the Production Cost Simulator (PROSIM) that dispatches the system companies on an hour-by-hour basis. In takes into account the characteristics of all units and entitlements available to the system companies, with regard to their heat rates, capabilities, seasonal deratings, forced outage rate, maintenance schedules, and minimum running and down times. The model accounts for capacity additions, retirements, or deratings that are assumed to occur during the period of analysis. All available units and entitlements are economically dispatched to meet the projected profile of system hourly load levels.

Cost parameters important in production simulation are also incorporated in PROSIM, such as variable operation and maintenance expenses, and fossil and nuclear fuel cost estimates. The resultant output of PROSIM is an estimate of the total monthly and annual fuel cost expense of the Northeast Utilities system companies.

PROSIM is a deterministic (versus probabilistic) model and will handle any electric utility system which includes conventional, standby and cold storage thermal units, nuclear units, and hydro units of the run of river, pondage, and pumped storage types. The program is used primarily as a planning device for production cost evaluations in the areas of capacity expansion planning, fuel budgeting, forecasting, and operation and maintenance evaluation.

First, PROSIM dispatches conventional hydro against the highest load level hours, recognizing the constraints of limited hydro energy available during each day of each month. Second, pumped hydro energy is dispatched on an economic basis in accordance with the pumping to generation cost ratios. This cost ratio is a function of the system's available thermal generating resources. The pumped hydro dispatch is solved using a weekly cycle, in accordance with the dispatch practice adopted by NEPOOL and predicated by the design size of the Northfield Mountain facility. Finally, thermal units are dispatched against the remaining hourly load levels on a priority costing basis, and with an appropriate consideration to their operational characteristics and limitations.

The complete PROSIM model consists of the following sub-routines or models, in addition to the main program which accesses these subroutines and controls the preparation of the output summary information: load model, operating reserve, overhaul schedule, equivalent forced outage rates, conventional hydro dispatch, pumped hydro dispatch, and thermal unit dispatch. The incremental costs of generation for the thermal units such as those at the WSGS differ for each plant. The incremental costs of generation are based on the operating characteristics of each plant and the associated operating and main-tenance costs.

If the WSGS is converted to coal, it will have a low incremental cost of operation and will become a base-loaded plant. The plant would be dispatched continuously at full load. As a result, the conversion of Units 1 and 2 for district heating would not impact on the hours of operation of the units. The impact of the district heating system would be that the electrical generation from the plant would be reduced and the lost power would have to be replaced by another plant at a cost equal to the incremental cost of power for the pool corresponding to the system load.

The analysis of the district heating system considers the cost for replacing this power as a penalty against the district heating system. The estimated electrical capacity reductions for WSGS Units 1 and 2 are shown in Tables 6-2 and 6-3.

If the WSGS was considered for conversion to district heating operation as a non-base-loaded plant, the analysis would be more complicated. A typical load duration curve for the operation of the plant as a pure electric plant would have to be developed. This load duration curve would then be compared with the load duration curve of the plant operating as a cogeneration plant. This analysis would probably show that there were some periods during which the district heating system reduced the electrical output of the units, and others when the district heating system required the station to run at a higher load than the pure electric case. Since the pure electric case represents the most efficient operation from the pool viewpoint, any change in operation would be a decrease in dispatch efficiency, and a penalty would be charged against the district heating system. This penalty would differ depending on the change in operation required. If the plant were required to decrease load, the penalty would be equal to the incremental fuel cost for the particular system load. If the plant were required to run at a high load, the penalty would be equal to the incremental cost associated with the WSGS.

Year	1985-1987	1988	1989	1990	1991	1992	1993
January	0	4.4	4.4	4.4	4.4	4.4	4.4
February		2.3	3.7	3.9	4.4	4.4	4.4
March		1.8	3.0	3.3	4.4	4.4	4.4
April	0	0	0.3	0.4	1.9	2.2	2.5
May	0	0	0	0	0	0	0
June	0	0	0	0	0	0	0
July	0	0.7	0.9	0.9	1.3	1.4	1.5
August	0	0	0	0	0	0	0
September	0	0	0	0	0	0	0
October	0	0	0.9	1.0	2.7	3.1	3.4
November	0	1.3	2.4	2.6	4.4	4.4	4.4
December	0	3.3	4.4	4.4	4.4	4.4	4.4

### Table 6-2

WSGS UNIT 1 ESTIMATED ELECTRICAL CAPACITY REDUCTION (MWe)

### Table 6-3

### WSGS UNIT 2 ESTIMATED ELECTRICAL CAPACITY REDUCTION (MWe)

	Year	1985-1990	1991	1992	1993
January		0	4.4	4.4	4.4
February		0	2.2	2.8	3.5
March		0	1.4	2.0	2.4
April		0	0	0	0
May		0	0	0	0
June		0	0	0	0
July		0	0	0	0
August		0	0	0	0
September		0	0	0	0
October		0	0	0	0
November		0	0.5	1.0	1.4
December			4.1	4.4	4.4

### Section 7 ENVIRONMENTAL ASSESSMENT

Installation of a district heating system in Springfield will cause the replacement of numerous uncontrolled small point sources with several large sources of emission. The latter will be more efficient than the individual units and will have control systems to reduce overall emissions.

The proposed district heating system will reduce the overall fuel consumption within the city because of the higher efficiencies of the central sources when compared to the individual point sources. The conversion of the WSGS to a cogenerating mode will increase its cycle efficiency and further reduce the overall fuel consumption within the city. It will also cause rejection of less heat to the environment per pound of fuel burned.

Elimination of area source emissions in downtown Springfield would, in and of itself, substantially improve air quality in the city. The net effect will depend upon changes in fuel use at WSGS due to the district heating system. At present, the station output cycles with demand. Serving a district heating system from this operating mode would increase emissions from the station, negating much of the air quality benefit gained from decreasing area source emissions. For the purposes of this study, it was assumed that WSGS would be converted to coal and run base loaded. It should be noted that coal conversion plans at the WSGS are uncertain at this time.

Assuming that the plant emissions remained unchanged, the impact of the district heating system would be a reduction in area point source emissions. The point sources in question are the individual residential, commercial and industrial customers that would convert to district heat supply. The energy consumption for the service area indicated the following fuel mix:

Fuel oil	83%
Coal	8%
Natural gas	8%
Electricity	1%

#### Table 7-1

### PROJECTED DISPLACED FUEL

Year	Annual Heat Consumption (MWht)	Displaced Oil Consumption (10' gal)	Displaced Coal Consumption (tons)	Net Displaced (1) Gas Consumption (10 <sup>7</sup> cu ft)
1985	76,490	2636	1392	-30.99
1986	76,490	2636	1392	-30.99
1987	132,762	4575	2416	-10.65
1988	164,508	5669	2994	5.37
1989	202,458	6976	3685	1.14
1990	211,897	7302	3857	-0.64
1991	290,804	10021	5293	4.89
1992	307,330	10590	5594	3.07
1993	322,157	11101	5864	0.53

(1) Adjusted for HTHW generator gas consumption

The reduction in emissions due to fuel displacement was calculated and assumed to be evenly spread across the service area. The projected displaced fuel for the development stages is shown in Table 7-1. When completed in 1993, the district heating system would have an annual load of approximately 322,160 MWht and displace about 11.1 million gallons of fuel oil and 5860 tons of coal per year. At the same time, the net natural gas consumption would decrease by about 5.3 million cubic feet per year. All gas used for district heating would be burned at substantially increased conversion efficiency.

Neither the Massachusetts Department of Environmental Quality Engineering nor the Lower Pioneer Valley Regional Planning Agency had sufficiently detailed information on fuel use by location to develop an accurate emission grid. Therefore, it was decided to calculate the emission reduction due to the total fuel displacement. The analysis was keyed to sulfur dioxide emissions and concentrations, since that pollutant will show the greatest incremental effect. Emission from natural gas firing was found to be negligible. Using emission factors from AP-42 (Compilation of Air Pollutant Emission Factors, U.A. EPA, 1980) it was determined that 11.1 million gallons of oil at 0.3% sulfur release 262.1 tons of sulfur dioxide. Additionally, 111.4 tons of sulfur dioxide would be released by burning 5860 tons of coal at 1% sulfur. The total sulfur dioxide emission reduction would be 373.5 tons per year. This emission reduction must be converted to ground level concentrations to estimate the air quality improvement. Several techniques are available to do this, most of which require details that are not available for the present sources of pollution. However, EPA offers one method which gives a reasonable approximation for concentrations within the study area and does not require detailed emission data. This method uses the equation:

$$C = 18 \frac{Q}{U} (\triangle X)^{1/4}$$

where:

C = maximum ground level concentration from area source  $(g/m^3)$ 

Q = average emission rate (g/m<sup>2</sup>/sec)

- U = average wind speed for period of concern (m/sec)
- $\Delta X$  = length of a side of the square that contains the bulk of emissions (m)

Most of the emissions in the service area can be represented by a square with 4X = 2500 m. Based on air quality measurements in the area, the wind speed on days with elevated sulfur dioxide levels averages 4-5 m/sec. Using these values in the above equation, the area source contribution to the high 24-hour sulfur dioxide averages would be  $89-115\mu \text{g/m}^3$ . Therefore, elimination of these area source emissions could reduce the maximum sulfur dioxide levels by that amount.

These air quality improvements assume that emissions from WSGS would not be affected by the district heating system. If emissions at the station were to increase, annual concentrations might actually go up. However, this would probably not be a problem with a "typical" district heating system. The WSGS is not well designed for dispersion, and strongly affects the air quality in downtown Springfield. The added dispersion from most power plants (with respect to replaced area sources) is likely to more than compensate for an increase in emissions.

7-3

### Section 8 INSTITUTIONAL ASSESSMENT

### OWNERSHIP OPTIONS AND FINANCIAL CONSIDERATIONS

#### Ownership Options

Many options are available for construction and operation of the SWERF and establishment of a district heating system in Springfield, each with advantages and disadvantages. The options selected for discussion herein are not exhaustive and, no doubt, alternatives could be suggested. Other combinations of entities beyond those considered are possible, but disadvantages would arise when departing from the single entity arrangement.

NORTHEAST UTILITIES OR A NEW SUBSIDIARY TO FINANCE, OWN AND OPERATE THE DISTRICT HEATING SYSTEM.

Representatives of Northeast Utilities indicated an interest in establishment of a district heating system in Springfield. The prospective district heating system may make it economically feasible to upgrade their WSGS to a base load operation by converting from oil to coal. However, even when firing coal the units may displace more efficient units when required to run to supply the district heating system, and penalty charges would be assessed to protect Northeast's electric customers. Northeast's present capital commitments would make it difficult to undertake a large new project with substantial funding required for district heating.

They indicated some interest in the concept of an unregulated subsidiary funded independently of their present capitalization. This possibility was mentioned in the most preliminary fashion. The restrictions on such a course have not been explored and could be formidable under the public utility laws of Massachusetts.

Because operation of a SWERF and administration of a waste disposal business differ so from running an electric utility company, Northeast Utilities is unlikely to participate in this segment of the project. Regardless of the final ownership, Northeast Utilities has assured the officials of Springfield that they will cooperate with the project to the fullest extent possible. The modification of the WSGS to provide hot water service for a district heating system will depend, however, on financial and contractual arrangements adequate to cover the investment, increased operating expenses, and incremental fuel costs.

The Massachusetts Legislature periodically considers a statute that would impose rate and other regulations on district heating utilities. Such a law would impose on Northeast Utilities an additional regulatory burden for district heating. Regardless of whether district heating rates are regulated, the allocation of capital, operating, and fuel costs between electric power and hot water production would impose a regulatory burden on Northeast Utilities in electric rate cases. Such regulatory burdens would not accrue to the city under municipal ownership and would not, or at least not to the same degree, be imposed on other optional ownerships.

This option has only a fair to poor chance of realization because of the reluctance of electric utilities to go into the district heating business.

SPRINGFIELD TO FINANCE, OWN, AND OPERATE THE DISTRICT HEATING SYSTEM, SELLING ELECTRICITY TO AND BUYING HOT WATER FROM NORTHEAST UTILITIES' WSGS

Officials of the Public Works Department said that the city is virtually precluded from undertaking the district heating project. Among the obstacles are the bonding and taxing limits imposed by Proposition 2-1/2 (passed in 1980); the delays inherent in complying with the state statutes that require competitive bidding for supplies, materials and replacement parts costing \$2,000; and the constraints of civil service employment rules that make the hiring, promoting and replacing of competent technical, construction, operating and maintenance personnel extremely difficult.

Another constraint on municipal ownership would be caused by the city's annual budget process. If the City Council in a given year should refuse to authorize adequate capital and operating funds, the extension of district heating service to new customers or the expansion of service to existing customers would be delayed. Even the continuing operation of the system could be put in jeopardy in case of budget crisis. Only minimal offsetting advantages would accrue to city ownership such as the right to use the streets and other public property for installation of facilities without the need for franchises or siting permits. Some economies might be possible by the use of the same maintenance and construction forces as are used for the water and sewer systems. Also, the profits from the sale of electricity to Northeast Utilities and heating service to consumers would supplement tax revenues. The city would benefit, however, under other entity options by reduced waste disposal expenses and possible franchise fees.

ENTREPRENEUR TO FINANCE, OWN AND OPERATE THE DISTRICT HEATING SYSTEM, INCLUDING THE HTHW BOILER PLANT AND SWERF, SELLING ELECTRICITY TO AND BUYING HOT WATER FROM NORTHEAST UTILITIES' WSGS.

Having an independent entrepreneur undertake the project would have a number of advantages. Under present law, an entity established for this purpose would be free of some of the constraints that Northeast Utilities and the city would encounter. An entrepreneur would be free to organize in one of several ways, including a general partnership with limited partners or a new or existing corporation. Likewise, he would be free to finance the facilities by a variety of means. The city has expressed willingness to cooperate by applying for HUD/UDAG grants and, if feasible, to devote some Block Grant funds to this purpose.

Officials of two large Springfield financial institutions indicated that they know of entrepreneurs who might be attracted to this project. They offered to assist in finding such a person or organization and in structuring the financial package.

In attracting an entrepreneur, the city would be offering assured revenues from waste tipping fees, and Northeast Utilities would buy the cogenerated electric power on a long-term contract.

The principal disadvantages of this option are the burdens an independent entity might have in building up a construction, operating and maintenance organization and the delay in the start of cash flow that would have to be covered in the original capitalization.

District heating projects under construction in Lawrence/Haverhill, MA, and

Trenton, NJ, are excellent examples of the entrepreneurial method of organizing a district heating enterprise. The Lawrence/Haverhill project includes waste recovery facilities, the Trenton project does not. However, the Trenton project has city and other government buildings as initial customers.

NON-PROFIT CORPORATION ESTABLISHED BY CITY AND COMMUNITY LEADERS TO FINANCE, OWN AND OPERATE THE DISTRICT HEATING SYSTEM, INCLUDING THE HTHW BOILER PLANT AND SWERF, SELLING ELECTRICITY TO AND BUYING HOT WATER FROM NORTHEAST UTILITIES' WSGS.

The concept of establishing a non-profit corporation to undertake the construction and operation of the district heating system has a number of advantages, although it may not be as easily carried out as the independent entrepreneurial option. The operating margin of a non-profit corporation need not be as great as that of the utility or an entrepreneur, so the tipping fees to the city and the heating rates to the consumers would probably be lower. With lower fees and rates, the activities of the two enterprises would probably grow more rapidly. With lower rates, district heating service would also compete with oil and gas heating more successfully.

To form a non-profit corporation to undertake this project, a major leadership effort would be required by government and community leaders. Also, substantial funds would be needed to cover organizing and planning expenses. Some preliminary planning has been done and, if a second HUD feasibility study grant is forthcoming, money would be available to cover some portions of these initial costs.

In St. Paul, MN, the non-profit corporation scheme has been successfully used to undertake a district heating project which is on a scale larger than that proposed for Springfield. Based on this, the non-profit corporation option would appear preferable to the others, except the entrepreneur. This option is heavily dependent on the willingness of the leaders of Springfield to expend the considerable effort required in its formation and initial funding. ENTREPRENEUR FINANCING, OWNING AND OPERATING THE SWERF, AND A NON-PROFIT CORPORATION FINANCING, OWNING AND OPERATING THE DISTRICT HEATING SYSTEM

Under this scheme the independent organization to fund, own, and operate the SWERF would have expertise in this field. This option would bring the SWERF on line in a shorter time. With a non-profit corporation funding, owning, and operating the district heating system, lower and more competitive heating rates might result. Under this option, the non-profit corporation would buy hot water from the SWERF operator and Northeast Utilities' WSGS.

For this option to succeed, it would be necessary to obtain substantial take-orpay consumer commitments initially so that revenue in early years would support the engineering and administrative manpower that would be required. Some additional difficulty would arise in having to bring into operation two new entities.

Springfield District Heating and Waste Recovery Authority to be established by the Massachusetts Legislature and provided with start-up funds to cover some of the planning, organizing and administration costs.

The authority concept has substantial appeal, particularly if it were to bring seed money to the project. It would have advantages similar to those of the non-profit corporation option. In addition, raising of capital funds might be more readily accomplished. However, it is improbable that the legislature would establish a new authority for the narrow purpose of waste recovery and district heating in Springfield alone. It is conceivable that a state-wide authority could be created, but from Springfield's point of view such an entity would have disadvantages in terms of delays and political complications. Also, it would probably take the legislature two or three annual sessions to bring such a proposition to a favorable vote.

NORTHEAST UTILITIES OR A NEW SUBSIDIARY COMPANY TO MANAGE AND OPERATE THE ENTIRE PROJECT UNDER CONTRACT WITH THE CITY OR AN INDEPENDENT ENTITY, INCLUDING OR EXCLUDING THE SWERF.

The concept of contracting with Northeast Utilities to manage the project would appear to have some attraction, since their Western Massachusetts Electric operating company is conveniently located and has substantial technical organization in place. It is doubtful, however, if the utility would view this idea

#### sewers, and utility facilities.

Major advantages of the UDAG program are flexibility and an expeditious review process. Funds are awarded to local governments, which can then lend or grant them to private or municipal developers. Flexibility in management of funds is designed to promote stronger working relationships among the local government, the commercial and industrial sectors, and the public in overcoming development problems. The rapid review process is another positive aspect of the program. Applicants can reasonably expect a decision on a proposal within two months of its submission. The review process is further enhanced by having four separate dates each year for filing applications.

Action Grant projects are selected on the basis of a national competition. A major eligibility criterion is a firm financial commitment from the private sector. The program is meant to catalyze increased investment in distressed communities by private sector involvement. Such investments must be firm before a grant can be approved.

Action Grants can be used for energy conservation or alternative energy supply projects, including district heating, which are technically feasible and have sufficient economic viability to attract the private investment required under the UDAG program. The minimum is \$2.5 private to \$1 Action Grant, but the higher the ratio of private investment to UDAG requested, the more competitive the application will be. As in other Action Grant applications, a partially guaranteed loan or a revenue bond (but not a local government general obligation bond) can qualify as private investment. Accordingly, the capitalization of any development entity must clearly be private in nature, with no obligation running entirely (100% guarantee) to a municipality, public authority or the general funds of a governmental entity in case of default of a project. The private investor must be at risk for the project, not the public participants, although public financial participation is not discouraged. Additionally, there should be new, permanent jobs created as a result of the project, although it is recognized that high technology, energy related projects do not create as many jobs as do some other investments.

An energy UDAG application would contain the following: Project Description - Provide a description of the community in terms of its existing and projected land use mix and zoning. The description should include a space heating market study of the building stock and related heating systems in the proposed service area. There should be maps that show the heat load density and significant detail on potential large customers within the service area. Finally, there should be a discussion of the utility service that exists within the area with specific emphasis on how the space heating load is served currently.

Development Plan - A time-phased integrated development plan for all phases of the project with key milestones identified would provide additional substantiation.

Engineering Design - The design should include all aspects of the district heating system, the piping system, the energy source and the conversion of the building heating systems. The expansion potential of the system without federal financing subsidies and the scarce fuel savings beyond the segment for which funding is sought should be estimated.

Economic Feasibility - Plans for financing the proposed district heating system should be outlined, including amounts, sources, and timing. The economic feasibility assessment should include all project costs and estimating assumptions, cash flow, capital amortization, and internal rate of return, both with and without the UDAG requested. The applicant should submit letters of intent from building owners to convert to the system at the energy prices used in the feasibility study as an indication of the project's economic viability. The applicant should also address the long term supply stability of the energy source of the proposed system.

Environmental and Legal - The project may be subject to local, county, or state environmental statutes and/or the national environmental protection laws. Provide an assessment of the applicable environmental regulations for the proposed system. Include in the development plan all task and key milestones denoting the successful completion of environmental-associated requirements. There should also be evidence that all legal requirements have been met or are scheduled and integrated into the development plan.

### REGULATORY AND LEGAL ASPECTS

### Introduction

This section provides a brief description of the laws and programs of the State of Massachusetts governing the regulation of public energy utilities and their probable direct or indirect impacts on the proposed Springfield project. To be considered are the siting of energy generating and transmission facilities, the municipal franchising of public energy utilities, and the prescription of rates to be charged by utilities, including attendant problems of cost allocations, rate base and operating expense determinations, and rate of return allowances.

The powers of regulatory agencies in most states are extensive and reach most aspects of the organization, operation and financing of the business of a public utility. Most regulatory commissions are granted supervisory authority over public utilities as well as specific powers. In addition to regulating rates, most commissions regulate the construction of utility facilities, transfer of assets and local franchises, initiation and abandonment of service, and standards of service. Also, they are empowered to regulate capitalization, issuance of securities, mergers and consolidations, and affiliated interest transactions and to prescribe a system of accounts to be followed by the utilities.

In examining the regulatory and legal aspects of district heating in Massachusetts, considerable guidance is provided in a report prepared for the U.S. Department of Energy by a Chicago law firm and issued in January 1981. This Massachusetts report, one of 50 state studies and an overview study, examines the laws of the state governing the regulation of public utilities, the siting of energy generating and transmission facilities, the municipal franchising of public utilities, and the regulation of rates.

#### Regulation of Public Utilities

The authority to regulate public utilities in Massachusetts is vested in the Department of Public Utilities (DPU), supervised by three commissioners appointed by the governor for 4-year terms. Municipalities are given a limited role in the regulation of public utilities. In general, the role of the municipality in Massachusetts is limited to the exercise of police power and to authorizing initiation of service by a new gas or electric company, if the municipality is already served by such a utility. The DPU is responsible for regulating and has supervisory authority over all gas and electric companies. The Massachusetts report does not mention district heating utilities as does the companion report on nearby Rhode Island where utility regulatory jurisdiction "...encompasses a wide variety of utility functions, including generation, manufacture, production, transmission, distribution or furnishing of natural or manufactured gas, steam, electrical or nuclear energy, heat, light or power."

### Siting of Energy Facilities

Massachusetts has enacted an electric power facilities siting law, and created the Energy Facilities Siting Council as a state agency responsible for implementing the siting law. The Council approves long-range forecasts for facilities which are required to be submitted by all gas and electric companies, and issues permits for the construction of such facilities after finding that the proposed construction is consistent with the approved long-range forecast. Upon application by a gas or electric company showing that it has been impeded in its efforts to obtain necessary permits from other state agencies or local governmental units, the Council may issue a certificate of environmental impact and public need which supersedes all other state and local restrictions that would delay or prevent the construction, operation or maintenance of the proposed facility.

Whether the Council and the siting law it administers would have jurisdiction over the SWERF and district heating system is not clear. The language of the law refers to electric and gas companies, so the question might arise, if a new district heating entity were formed that would generate electricity in the SWERF whether it would be considered an electric utility subject to the law.

### Franchising of Public Utilities

The authority to grant franchises for the use of public streets and places in connection with providing utility services in Massachusetts is vested in municipalities. To be granted a franchise, the statutes require that a public utility must file a petition with the local governing body, after which a public hearing must be held. There are statutory criteria governing whether or not a franchise should be awarded.

Specifically, municipalities are empowered to "authorize the laying of pipes and conduits for the transmission of steam or hot water for heating, cooling and mechanical power, for private use..." A separate statutory provision states

specifically that corporations organized for the purpose of generating and furnishing steam or hot water may lay pipeline only with the written consent of the local authorities.

Franchising authority is not limited to certain classes of cities. The statutes do not limit the type of entities which may be franchised, but the franchising power is limited to entities providing electricity, steam, hot water or other enumerated services.

Certain procedures have been established by statute governing the grants of franchises for the construction of electric facilities, but no specific procedures have been established with respect to other franchise grants. Local home rule charters and ordinances must be consulted to determine the procedure in obtaining a franchise.

No specific criteria for evaluating franchise requests have been established by statute or reported judicial decision. A franchisee need not obtain a certificate of convenience and necessity from the state before seeking a franchise, and franchises need not be awarded competitively to the highest bidder. No maximum term for a franchise has been established by statute or by judicial decision, nor are perpetual franchises specifically sanctioned or prohibited. Municipalities have been neither authorized nor forbidden to grant exclusive franchises by statute or reported judicial decision.

### Rate Regulation

Currently, district heating rates are not regulated by the DPU in Massachusetts. However, there is the ever-present possibility that the Legislature may extend the regulation of rates to steam and hot water service that now applies to electric and gas services. Such regulation of district heating does occur in Rhode Island and New York.

The regulation of electric and gas service requires that no change in rates can be made until an amended rate schedule is filed, hearings are held, and a decision is made by the DPU. Rate changes may be suspended for a period not to exceed six months while the hearing procedures and submissions of briefs and reply briefs are carried out and the commissioners weigh the evidence before reaching a decision. In relation to rate regulation, the utility is required to keep financial records according to a uniform system of accounts. In Massachusetts the Federal Energy Regulatory Commission (FERC) system of accounts is employed. Also, the DPU may order a utility to establish and maintain a depreciation reserve fund that it deems adequate.

The Massachusetts statutes give the DPU authority to "supervise" the rates of certain municipally owned utilities. Rates charged by a municipality may not be fixed at less than production cost as determined by the DPU and may not yield more than eight percent per year, after operating cost, interest, sinking fund provision and depreciation. Inherent in rate regulation are the DPU policies and practices relating to rate base determination, valuation of property, test periods, allowed expenses, treatment of amounts in construction work in progress, and other related accounting and financial matters.

Of overriding concern under rate regulation is the method used by the DPU for determining the rate of return to be allowed a regulated utility. The Massachusetts DPU, in specifying the overall rate of return in a given case, focuses heavily on the cost of equity as well as the cost of service and finds that the discounted cash-flow method of determining reasonableness is preferable. Other factors considered in determining rate of return allowed include the utility capital structure, risk, quality of service, efficiency of management and nature of operations.

The regulatory climate for utilities in Massachusetts could, of course, be examined herein in far greater depth. Until such time as the DPU is empowered to regulate district heating utilities, this brief overview may suffice.

Under present Massachusetts law which does not provide for the regulation of rates for steam or hot water, the sale of heat by Northeast Utilities to the district heating entity would not be subject to state rate regulation, as such. Northeast Utilities, however, would encounter a measure of regulation in electric rate cases when the method and reasonableness of the allocation of operating and fuel costs between electric generation and steam or hot water for district heating at the WSGS is examined. Also, it is likely that a contract for heat between Northeast Utilities and the district heating entity would require filing with and approval by the DPU. As an example of the indirect regulation of district heating, Boston Edison reports that in past years when their unregulated street steam system was supplied from non-condensing electric turbines, the question of cost allocation always arose in electric rate cases, with intervenors on behalf of both the steam customers and the electric customers challenging the utility's computations. Boston Edison also reports that for the past several years, at the instigation of the Building Owners and Managers Association of Boston, legislation has been proposed that would authorize the DPU to regulate district heating. Each year it has been defeated in committee.

Another district heating system now operating in Massachusetts is Cambridge Electric Light Company's comparatively small system. Of their 41 steam customers, Harvard University, that owns and maintains its own piping in the streets from the campus to the power plant, is the largest. The steam is supplied from cogeneration sources.

In Lawrence, construction has been started on a district heating system that will include a waste recovery plant, a cogeneration power plant and a district heating system. The heat distribution medium will be steam which will be supplied initially to several industries and two housing projects. Most of the electric power will be sold to New England Power Company under PURPA rates. Some small electricity sales will continue to be made within the Arlington Mills Complex, an industrial park where the power plant is located. The understanding of the management of the Lawrence project is that the sale of steam will not be subject to rate regulation. They recognize, however, that under applicable provisions of Massachusetts law, retail sales of electricity may be subject to retail rate regulation, but they do not believe their project will be subject to DPU financial or organizational regulation. The DPU, under Chapters 164 and 94, has the authority to regulate electric rates if a cogenerator or district heating entity engages in retail sales of electricity, as will occur in Lawrence.

### Federal Regulation

Section 210(e) of the PURPA Act of 1978 provides the FERC with authority to promulgate regulations exempting certain cogeneration and small power production facilities from various state and federal public utility regulatory laws. Under the regulations promulgated by FERC under Section 210, cogeneration and small power production facilities satisfying certain specified criteria ("qualifying facilities") are exempt from regulation under the Public Utility Holding Company Act of 1935, wholesale rate regulation under the Federal Power Act, as amended, and certain Massachusetts regulatory laws.

The Springfield SWERF cogenerating plant will be owned and operated by an entity other than Northeast Utilities, and, therefore, will be considered a qualifying facility under PURPA and be exempt from the forms of public utility regulation mentioned above. The sale of the cogenerated electric power to Northeast Utilities would be subject to PURPA regulations promulgated by the Massachusetts DPU and the Western Massachusetts Electric Company's filed Power Purchase Schedule tariff MDPU 477. Under the provisions of MDPU 477, the owner of the SWERF would be paid for electricity furnished to the utility in accordance with avoided cost on an on-peak and off-peak hours basis. The avoided costs are recomputed quarterly. For the December 1982 through February 1983 they were:

On-peak	primary	5.35¢/kWh
On-peak	secondary	5.57¢/kWh
Off-peak	primary	4.00¢/kWh
Off-peak	secondary	4.16¢/kWh

The MDPU 477 filed by Western Massachusetts Electric as required by the DPU provides a minimum range for the pricing of sale of power from the project owner to the utility. Nothing in the PURPA rules, however, precludes the cogenerating entity and the utility from negotiating a mutually satisfactory agreement outside the scope of the filed buy-back tariff.

The annual carrying charges were based on the assumption that the entire lists would be owned by a municipal-type entity and that heat would be purchased into the MSGS and the SWERF. The project mould be financed using 100% debt except for the modifications of the MSGS (phases 30 and 68). The cost of monitival debt is 71. The income tax and property tax rates for a municipality are of, and the insurance rate is 0.51. The analysis is conducted for a 30-year book life.

The annual carrying charges for phases 30 and 60 were based on the assumption that Northeast Utilities would endify their WSGS to sell district heat to the punicipality at a rate engal to the carrying charges for the modification plus
## Section 9 ECONOMIC ANALYSIS

The economic analysis of hot water district heating in Springfield was performed from the viewpoint of municipal ownership except for the modification of the Northeast Utilities' WSGS. The analysis determined the annual carrying charges for each investment phase, the composite carrying charges for all investment phases, the unit cost of district heat, and the pay-back period for specific customer retrofit categories.

The analysis uses the required revenue approach to determine the necessary charges for district heating. The annual carrying charges for the district heating investment are calculated based on a debt to equity ratio, return rates, book life, tax information, and insurance rates, both typical of municipal ownership and for Northeast Utilities. The method was used to develop the total system costs and compare these costs with the total quantity of heat sold to determine the minimum required charges for district heat. The total system costs are comprised of fixed expenses, operating expenses, replacement electricity, and gross receipt taxes.

The capital costs were derived in 1983 dollars and then escalated to the year of expenditure at a rate of 6.5%. The capital costs include all direct and indirect costs associated with the power plant retrofits and the transmission and distribution systems.

The annual carrying charges were based on the assumption that the entire system would be owned by a municipal-type entity and that heat would be purchased from the WSGS and the SWERF. The project would be financed using 100% debt except for the modifications of the WSGS (phases 3B and 6B). The cost of municipal debt is 7%. The income tax and property tax rates for a municipality are 0%, and the insurance rate is 0.5%. The analysis is conducted for a 30-year book life.

The annual carrying charges for phases 3B and 6B were based on the assumption that Northeast Utilities would modify their WSGS to sell district heat to the municipality at a rate equal to the carrying charges for the modification plus the replacement electricity costs. The plant modification is assumed to be financed with a weighted cost of capital equal to 14.38%. The income tax rate for the utility is 46%, the insurance rate is 0.5%, and the gross receipts tax is 4%. The utility is assumed to take advantage of accelerated tax depreciation on the modification over a 15-year period.

The annual carrying charges for the eight phases of implementation are presented in Tables 9-1 through 9-10; the composite annual carrying charges are presented in Table 9-11.

The operating expenses for the district heating system are comprised of replacement electricity costs, fuel costs, refuse heat costs, pumping costs, operating and maintenance manpower, and operating and maintenance materials. The replacement electricity costs are charged against the district heating system to compensate for the reduction in electrical output at the WSGS caused by the district heating modification. The replacement electricity costs are \$51.44/MWh in 1983 dollars. The fuel costs charged to the system represent the cost of natural gas burned in the high temperature hot water boilers. The gas costs are \$5.13/MBtu in 1983 dollars. The refuse heat costs in the analysis are for heat purchased from the SWERF. The cost of refuse heat is \$4.81 MWt in 1983 dollars. Both gas and refuse heat costs are escalated at 7.5%. Pumping costs for transmission, distribution costs, and condensate return are also calculated using \$51.44/MWh in 1983 dollars. Electric power costs are escalated at 7.5% per year. Operating and maintenance manpower for the system is estimated to cost \$35,000 per man-year in 1983 dollars, including overhead and benefits, escalated at an annual rate of 6.5%. Operating and maintenance material costs are estimated to be equal to 3% of the capital costs of the heat source and 1% of the capital costs of the piping on an annual basis, escalated at 6.5%. The quantities of replacement electricity and pumping power were determined from the load duration curves for the respective phases of development.

The economic analysis developing the cost of district heating is presented in Table 9-12; Figure 9-1 shows item H graphically.

An analysis to determine the period in which commercial heating customers would recover their retrofit expenses was performed for two different cases and several different original building heating systems. One case assumes the customer does not finance the retrofit, the other that he finances the retrofit at an annual rate of 12% for 20 years. The equipment is depreciated over a 5year period using an accelerated depreciation and an expensing deduction in the first year. Customer maintenance and repair costs for the district heating equipment are not included. Maintenance expenses should be lower for district heating equipment than for existing heating equipment. The payback period is based on the after tax savings in annual energy costs. This analysis assumes a tax rate of 50%. The analysis was performed for an energy escalation rate for oil of 7.5%; it was determined that over 80% of the existing heating systems in Springfield burn oil.

The unit cost of district heat is compared with the unit cost of oil, adjusted for combustion efficiency, in Figure 9-2. The consumer payback periods for different types of conversions are developed in Tables 9-13 through 9-17 for 7.5% oil escalation. The payback period depends on the type of existing system and varies form 3 to 6 years for most consumers.

Results of the economic analysis demonstrate that a hot water district heating system will supply heat at lower cost than individual boilers fired with oil.





Figure 9-2. Projected Heating Cost Comparison

9-4

# ANNUAL CARRYING CHARGES, PHASE ONE

Investment for Phase	1					Econ	omic Facto	rs		100	198672045
Initial Operation	1985	Pr	eferred St	ock Ratio		0	B	ook Life - '	Yrs		30
Cost Racis	1983	Re	turn on Pr	eferred S	tock - %	0	I	ncome Tax R	ate - %		0
Canital Cost - \$1000		Co	mmon Stock	Ratio		0	ī	ax Credit -	7.		0
Heat Gource	1408	Re	turn on Co	amon Stoc	k - %	0	T	ax Life - Y	15		30
Pining	4347	De	bt Ratio			1	A	ccelerated	Tax Depre	ciation	0
Total	5755	De	bt Cost -	%		7	P	roperty Tax	- %		0
Feralation - \$1000	570	We	ighted Co	st of Capi	tal - %	7	Interne To	nsurance Ra	te - %		0.5
AEDC - \$1000	218										
Total Cost - \$1000	6543										
Start of Evaluation	1985									22.010.07	1
Year		1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
Presiding Bank Uslup		4543	6325	6107	5888	5670	5452	5234	5016	4798	4580
Remaining Book value	hast	0343	0010	0	0	0	0	0	0	0	0
Return on Freterred S	LULK		0	0	0	0	0	0	0	0	0
Keturn on Lowmon Stor	. K	458	243	427	412	397	382	366	351	336	321
Interest Deck December		210	218	218	218	218	218	218	218	218	218
BOOK Depreciation		210	210	218	218	218	218	218	218	218	218
lax vepreciation		210	210	0	0	0	0	0	0	0	0
Income laxes		0	0	0	0	0	0	0	0	0	0
Deterred Taxes		0	0	0	0	0	0	0	0	C	0
Property lax		77	77	रर	77	33	33	33	33	33	- 33
Insurance		33	33	55	55						
Annual Carrying Charg	ges	709	694	678	663	648	632	617	602	587	571
Year		1995	1996	1997	1998	1999	2000	2001	2002	2003	2004
Panaining Pook Value		4362	4144	3926	3708	3489	3271	3053	2835	2617	2399
Poturn on Proferred	Stock	0	0	0	0	0	0	0	0	0	- 0
Peturn on Compon Sto	rk	0	0	0	0	0	0	0	0	0	0
lotorest		305	290	275	260	244	229	214	198	183	168
Pook Depreciation		218	218	218	218	218	218	218	218	218	218
Tay Depreciation		218	218	218	218	218	218	218	218	218	218
Income Tayes		0	0	0	0	0	0	0	0	0	0
Deferred Tayas		0	0	0	0	0	0	0	0	0	0
Property Tax		0	0	0	0	0	0	0	0	0	0
Tocurance		77	77	33	33	33	33	23	33	22	33
insur dire		55	50							474	
Annual Carrying Char	nes.	556	541	526	510	495	480	465	449	454	419

### ANNUAL CARRYING CHARGES, PHASE TWO

Investment for Phase	2					Econo	mic Facto	rs			
Initial Operation	1987	Pr	eferred S	tock Ratio	)	0	8	ook Life -	Yrs		30
Cost Basis	1983	Re	turn on P	referred S	stock - %	0	I	ncome Tax R	ate - %		0
Canital Cost - \$1000		Co	anon Stoc	k Ratio		0	T	ax Credit -	7.		0
Heat Source	957	Re	turn on C	ommon Stoc	k - %	0	T	ax Life - Y	rs		30
Piping	2791	De	bt Ratio			1	A	ccelerated	Tax Depre	ciation	0
Total	3748	De	bt Cost -	. %		7	P	roperty Tax	- %		0
Escalation - \$1000	924	We	ighted Co	st of Capi	tal - %	7	I	nsurance Ra	te - %		0.5
AFDC - \$1000	161										
Total Cost - \$1000	4833										
Start of Evaluation	1985										
Year		1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
Remaining Book Value	92	0	0	4833	4672	4511	4350	4189	4027	3866	3705
Return on Preferred Stoc	k	0	0	0	0	0	0	0	0	0	0
Return on Common Stock		0	0	0	0	0	0	0	0	0	0
Interest		0	0	338	327	316	304	293	282	271	259
Book Depreciation		0	0	161	161	161	161	161	161	161	161
Tax Depreciation		0	0	161	161	161	161	161	161	161	161
Income Taxes		0	0	0	0	0	0	0	0	0	0
Deferred Taxes		0	0	0	0	0	0	0	0	0	0
Property Tax		0	0	0	0	0	0	0	0	0	0
Insurance		0	0	24	24	24	24	24	24	24	24
Annual Carrying Charges		0	0	524	512	501	490	478	467	456	445
Year		1995	1996	1997	1998	1999	2000	2001	2002	2003	2004
Remaining Book Value		3544	3383	3222	3061	2900	2739	2578	2416	2255	2094
Return on Preferred Stor	ck	0	0	0	0	0	0	0	0	0	0
Return on Compan Stock		0	0	0	0	0	0	0	0	0	0
Interest		248	237	226	214	203	192	180	169	158	147
Book Depreciation		161	161	161	161	161	161	161	161	161	161
Tax Depreciation		161	161	161	161	161	161	161	151	161	161
Income Taxes		0	0	0	0	0	0	0	0	0	0
Deferred Taxes		0	0	0	0	0	0	0	0	0	0
Property Tax		0	0	0	0	0	0	0	. 0	0	0
Insurance		24	24	24	24	24	24	24	24	24	24
Annual Carrying Charges		433	422	411	400	388	377	366	354	343	332

9-6

-

## ANNUAL CARRYING CHARGES, PHASE THREE A (MUNICIPAL)

Investment for Phase	3 A					Eco	nomi	c Factor	rs	1011		eless in
Initial Operation 19	88	Pref	erred Sto	ck Ratio	in toold	0		B	ook Life -	Yrs		30
Cost Basis 19	83	Retu	irn on Pre	ferred Stoc	k - %	0		I	ncome Tax	Rate - %		0
Capital Cost - \$1000		Com	ion Stock	Ratio		0		T	ax Credit	- %		0
Heat Source	0	Retu	irn on Com	mon Stock -	7.	0		T	ax Life -	Yrs		30
Pining 15	583	Debt	Ratio			1		A	ccelerated	Tax Depri	eciation	0
Total 15	83	Debt	Cost - Z			7		P	roperty Ta	x - %		0
Escalation - \$1000	519	Weid	hted Cost	of Capital	- %	7		I	nsurance R	ate - %		0.5
AFDC - \$1000	72											
Total Cost - \$1000 21	74											
Start of Evaluation 19	85											
Year		1985	1986	1987	1988	1989		1990	1991	1992	1993	1994
Remaining Book Value		0	0	0	2174	2101		2029	1957	1884	1812	1739
Return on Preferred Stock		0	0	0	0	0		0	0	0	0	0
Return on Coason Stock		0	0	0	0	0		0	0	0	0	0
Interest		0	0	0	152	147		142	137	132	127	122
Book Depreciation		0	0	0	72	72		72	72	72	72	72
Tax Depreciation		0	0	0	72	72		72	72	72	72	72
Income Taxes		0	0	0	0	0		0	0	0	0	0
Deferred Taxes		0	0	0	0	0		0	0	0	0	0
Property Tax		0	0	0	0	0		0	0	0	0	0
Insurance		0	0	0	11	11		11	11	11	11	11
Annual Carrying Charges		0	0	0	236	230		225	220	215	210	205
Year		1995	1996	1997	1998	1999		2000	2001	2002	2003	2004
Remaining Book Value		1667	1594	1522	1449	1377		1304	1232	1159	1087	1015
Return on Preferred Stock		0	0	0	0	0		0	0	0	0	(
Return on Common Stock		0	0	0	0	0		0	0	0	0	(
Interest		117	112	107	101	96		91	86	81	76	71
Book Depreciation		72	72	72	72	72		72	72	72	72	7:
Tax Depreciation		72	72	72	72	72		72	72	72	72	7:
Income Taxes		0	0	0	0	0		0	0	0	0	
Deferred Taxes		0	0	0	0	0		0	0	0	0	1
Property Tax		0	0	0	0	0		0	0	0	0	AT LUCK
Insurance		11	11	11	11	11		11	11	11	11	1
Annual Carrying Charges		200	195	190	185	180		175	170	164	159	15

# ANNUAL CARRYING CHARGES, PHASE THREE B (UTILITY)

Investment for Phase	3B		Economic Factors											
Initial Operation	1988	Pr	eferred St	ock Ratio		0.11	Bo	ok Life -	Yrs		30			
Cost Basis	1983	Re	turn on Pr	eferred St	ock - Z	11	In	come Tax R	ate - Z		46			
Capital Cost - \$1000		Ca	ason Stock	Ratio		0.4	Ta	x Credit -	7.		10			
Heat Source	2216	Re	turn on Co	amon Stock	- %	17	Ta	1x Life - Y	r5		15			
Piping	0	De	bt Ratio			0.49	Ac	celerated	Tax Deprec	iation	1			
Total	2216	De	bt Cost -	Z		13	Gr	oss Receip	ts Tax - %		4			
Escalation - \$1000	726	We	righted Cos	it of Capit	al - %	14.38	Ir	isurance Ra	ite - %		0.5			
AFDC - \$1000	204													
Total Cost - \$1000	3146													
Start of Evaluation	1985					1 10 3000								
Year		1985	1986	1987	1988	1989	1990	1991	1992	1993	1994			
Remaining Book Value		0	0	0	2832	2689	2463	2249	2049	1861	1686			
Return on Preferred St	ock	0	0	0	34	33	30	27	25	23	20			
Return on Common Stock		0	0	0	193	183	167	153	139	127	115			
Interest		0	0	0	180	171	157	143	131	119	107			
Book Depreciation		0	0	0	94	94	94	94	94	94	94			
Tax Depreciation		0	0	0	210	392	364	336	308	280	252			
Income Taxes		0	0	0	136	43	40	38	38	38	39			
Deferred Taxes		0	0	0	48	132	119	106	93	80	68			
Property Tax		0	0	0	126	129	132	136	139	142	146			
Insurance		0	0	0	16	16	16	16	16	16	16			
Annual Carrying Charge	5	0	0	0	827	800	756	714	675	638	605			
Year	8	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004			
Remaining Book Value		1525	1375	1239	1116	1005	908	823	751	692	646			
Return on Preferred St	ock	18	17	15	14	12	11	10	9	8	8			
Return on Common Stock		104	94	84	76	68	62	56	51	47	44			
Interest		97	88	79	71	64	58	52	48	44	41			
Book Depreciation		94	94	94	94	94	94	94	94	94	94			
Tax Depreciation		224	196	168	140	112	84	56	28	0	0			
Income Taxes		40	43	47	51	56	63	70	78	87	83			
Deferred Taxes		55	42	29	16	3	-10	-23	-35	-48	-48			
Property Tax		150	153	157	161	165	169	173	178	182	187			
Insurance		16	16	16	16	16	16	16	16	16	16			
Annual Carrying Charge	25	574	546	521	499	480	463	449	438	430	425			

## ANNUAL CARRYING CHARGES, PHASE FOUR

Investment for Phase	4						Economi	c Factors				
Initial Operation Cost Basis Capital Cost - \$1000 Heat Source Piping Total Escalation - \$1000 AFDC - \$1000 Total Cost - \$1000 Start of Evaluation	1989 1983 470 1061 1531 634 74 2239 1985	kali kalis tian tian kati tian tian	Pref Retu Come Retu Debt Debt Weig	erred Stoc rn on Pref on Stock F rn on Comm Ratio Cost - % hted Cost	k Ratio erred Stoc Natio Non Stock - of Capital	k - % % - %	0 0 0 1 7 7	Boo Inc Tax Tax Acc Pro Ins	k Life - ome Tax R Credit - Life - Y elerated perty Tax urance Ra	Yrs ate - % % rs Tax Depre - % ite - %	eciation	30 0 30 0 0 0.5
Year	(79)	1981	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
Remaining Rook Value			0	0	0	0	2239	2165	2090	2015	1941	1866
Raturn on Proferrad C	tark		0	0	0	0	0 0	0	0	0	0	0
Return on Common Stor	·		0	0	0	0	0	0	0	0	0	0
Internet			0	0	0	0	157	152	146	141	136	131
Rook Denreciation			0	0	0	0	75	75	75	75	75	75
Tay Depreciation			0	0	0	0	75	75	75	75	75	75
Income Taxos			0	0	0	0	0	0	0	0	0	0
Deferred Tayor			0	0	0	0	0	0	0	0	0	0
Proparty Tay			0	0	0	0	0	0	0	0	0	. 0
Insurance			0	0	0	0	11	11	11	11	11	11
Annual Carrying Char	ges		0	0	0	0	243	237	232	227	222	216
Year			1995	1996	1997	1998	1999	2000	2001	2002	2003	2004
Remaining Book Value		139.	1791	1717	1642	1567	1493	1418	1344	1269	1194	1120
Return on Preferred	Stock		0	0	0	0	0	0	0	0	0	0
Return on Common Sto	ck		0	0	0	0	0	0	0	0	0	0
Interest			125	120	115	110	104	99	94	89	84	78
Book Depreciation			75	75	75	75	75	75	75	75	75	75
Tax Depreciation			75	75	75	75	75	75	75	75	75	75
Income Taxes			0	0	0	0	0	0	0	0	0	0
Deferred Taxes			0	0	0	0	0	0	0	0	0	0
Property Tax			0	0	0	0	0	0	0	0	0	0
Insurance			11	11	11	11	11	11	11	11	11	11
Annual Carrying Char	ges		211	206	201	196	190	185	180	175	169	164

### ANNUAL CARRYING CHARGES, PHASE FIVE

Investment for Phas	5							Ecor	ionic Fa	tors		881	et set to	independ	
Initial Operation Cost Basis Capital Cost - \$100 Heat Source Piping Total Escalation - \$1000 AFDC - \$1000 Total Cost - \$1000	1 100	990 983 470 636 106 559 57 723 985		Pr Re Ca Re De De	efer turn amon turn bt R bt C ight	red Stock on Prefe Stock Ri on Comm Ratio Cost - X ced Cost	k Ratio erred Stoc atio on Stock - of Capital	k - % % - %	0 0 0 1 7 7	in resident in Frederik in Kanadi in	Book Incor Tax ( Tax ( Acce Propr Insur	Life - Tredit Life - Lerated erty Ta rance R	Yrs Rate - % - % Yrs Tax Deprec: x - % ate - %	iation	30 0 30 0 0.5
Year	1442		19	85	1	1986	1987	1988	1989	199	0	1991	1992	1993	1994
Remaining Book Valu	ue			0		0	0	0	0	172	3	1665	1608	1550	1493
Return on Preferred	d Stock			0		0	0	0	0		0	0	0	0	0
Return on Common St	tork			0		0	0	0	0		0	0	0	0	0
Interest	LUCK			0		0	0	0	0	12	1	117	113	109	105
Rook Depreciation				0		0	0	0	0	5	7	57	57	57	57
Tax Depreciation				0		0	0	0	0	5	7	57	57	57	57
Income Taxes				0		0	0	0	0		0	0	0	0	0
Deferred Taxes				0		0	0	0	0		0	0	0	0	0
Property Tax				0		0	0	0	0		0	0	0	0	0
Insurance				0		0	0	0	0		9	9	9	9	9
Annual Carrying Cha	arges			0		0	0	0	0	18	7	183	179	175	171
Year	1996		19	95		1996	1997	1998	1999	200	0	2001	2002	2003	2004
Remaining Book Val	ue		14	36		1378	1321	1263	1206	114	8	1091	1034	976	919
Return on Preferre	d Stock			0		0	0	0	0		0	0	0	0	0
Return on Common S	tock			0		0	0	0	0		0	0	0	0	0
Interest				100		96	92	88	84	8	10	76	72	58	64
Book Depreciation				57		57	57	57	57	5	7	57	57	57	57
Tax Depreciation				57		57	57	57	57		7	57	57	57	57
Income Taxes				0		0	0	0	0		0	0	0	0	0
Deferred Taxes				0		0	0	0	0		0	0	0	0	0
Property Tax				0		0	0	0	0		0	0	0	0	0
Insurance				9		9	9	9	9		9	9	9	9	9
Annual Carrying Ch	arges			167		163	158	154	150	14	16	142	138	134	130

Investment for Phase	6 A	enotas? :	Contract.			Econom	ic Factors		88		
Initial Operation Cost Basis Capital Cost - \$1000 Heat Source Piping Total Escalation - \$1000 AFDC - \$1000 Total Cost - \$1000 Start of Evaluation	1991 1983 0 1855 1855 1120 102 3077 1985	Pre Retu Com Retu Deb Neit	ferred Stor urn on Pre- non Stock I urn on Com t Ratio t Cost - % ghted Cost	ck Ratio ferred Sto Ratio mon Stock of Capita	ck - % - % 1 - %	0 0 0 1 7 7	Booi Inc: Tax Tax Acc: Proj Insi	k Life - ome Tax R Credit - Life - Y elerated perty Tax urance Ra	Yrs tate - X 7 Tax Depr - X tte - X	eciation	30 0 30 0 0 0 0.5
Year	1171	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
Remaining Book Value		0	0	0	0	0	0	3077	2975	2872	2769
Return on Preferred Stor	k	0	0	0	0	0	0	0	0	0	0
Return on Common Stock		0	0	0	0	0	0	0	0	0	0
Interest		0	0	0	0	0	0	215	208	201	194
Book Depreciation		0	0	0	0	0	0	103	103	103	103
Tax Depreciation		0	0	0	0	0	0	103	103	103	103
Income Taxes		0	0	0	0	0	0	0	0	0	0
Deferred Taxes		0	0	0	0	0	0	0	0	0	0
Property Tax		0	0	0	0	0	0	0	0	0	0
Insurance		0	0	0	0	0	0	15	15	15	15
Annual Carrying Charges		0	0	0	0	0	0	333	326	319	312
Year	7005	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004
Remaining Book Value		2667	2564	2462	2359	2257	2154	2051	1949	1846	1744
Return on Preferred Stoc	k	0	0	0	0	0	0	0	0	0	0
Return on Common Stock		0	0	0	0	0	0	0	0	0	0
Interest		187	180	172	165	158	151	144	136	129	122
Book Depreciation		103	103	103	103	103	103	103	103	103	103
Tax Depreciation		103	103	103	103	103	103	103	103	103	103
Income Taxes		0	0	0	0	0	0	0	0	0	0
Deferred Taxes		0	0	0	0	0	0	0	0	0	0
Property Tax		0	0	0	0	0	0	0	0	0	0
Insurance		15	15	15	15	15	15	15	15	15	15
Annual Carrying Charges		305	297	290	283	276	269	262	254	247	240

# ANNUAL CARRYING CHARGES, PHASE SIX A (MUNICIPAL)

# ANNUAL CARRYING CHARGES, PHASE SIX B (UTILITY)

Investment for Phase 6B Initial Operation 1991 Preferred Stock Ratio					Econ	nomic Fact	tors		19 101 2000	Jasen I.		
Initial Operat	tion	1991	Pre	eferred Sto	ck Ratio	e idadi	0.11	Professo	Book Life -	Yrs		30
Cost Basis		1983	Ret	urn on Pre	ferred Sto	ck - Z	11		Income Tax I	Rate - Z		46
Capital Cost -	- \$1000		Con	mon Stock	Ratio	10	0.4		Tax Credit	- 7.		10
Heat Source	e	2216	Ret	urn on Com	non Stock	- %	17		Tax Life -	rs		15
Piping		0	Det	t Ratio			0.49		Accelerated	Tax Depr	eclation	1
Intal	*****	1770	Ver	It LOST - L			13		bross Recei	sts lax -	7	4
Escalation - a	\$1000	1338	#e1	gntea Lost	of Lapita	1 - 1	14.38		Insurance K	ate - 7		0.5
HFUL - \$1000	*1000	297										
Chart of Evalu	*1000	1005										
Start Of Evalu												
Year	r	1991	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
Remaining Book	k Value		0	0	0	0	0	0	3421	3248	2975	2717
Return on Pret	ferred Stock		0	0	0	0	0	0	41	39	36	33
Return on Coas	mon Stock		0	0	0	0	0	0	233	221	202	185
Interest			0	0	0	0	0	0	218	207	190	173
Book Depreciat	tion		0	0	0	0	0	0	114	114	114	114
Tax Depreciati	ion		0	0	0	0	0	0	253	473	439	405
Income Taxes			0	0	0	0	0	0	164	52	48	46
Deferred Taxes	s		0	0	0	0	0	0	58	159	144	128
Property Tax			0	0	0	0	0	0	152	156	160	164
Insurance			0	0	0	0	0	0	19	19	19	19
Annual Carryin	ng Charges		0	0	0	0	0	0	1000	967	913	862
Year	r Cont	1962	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004
Remaining Book	k Value	150	2475	2248	2037	1842	1662	1497	1348	1215	1097	994
Return on Pre-	ferred Stock		30	27	25	22	20	18	16	15	13	12
Return on Coa	non Stock		168	153	139	125	113	102	92	83	75	68
Interest			158	143	130	117	106	95	86	77	70	63
Book Depreciat	tion		114	114	114	114	114	114	114	114	114	114
Tax Depreciat:	ion		372	338	304	270	236	203	169	135	101	68
Income Taxes			45	45	47	49	52	56	62	68	76	84
Deferred Taxes	5		113	97	82	66	51	35	19	4	-12	-27
Property Tax			168	172	176	181	185	190	195	199	204	210
Insurance			19	19	19	19	19	19	19	19	19	19
Annual Carryin	ng Charges		815	771	730	693	660	630	603	579	559	543

# ANNUAL CARRYING CHARGES, PHASE SEVEN

E is Ester

Investment for Phase	7	ECONOMIC FACTORS										
Initial Operation Cost Basis Capital Cost - \$1000 Heat Source Piping Total Escalation - \$1000 AFDC - \$1000	1992 1983 0 741 741 525 44	Pre Ret Com Ret Det Wei	eferred Sto nurn on Pre nurn on Stock nurn on Coar of Ratio of Cost - 7 ighted Cost	ock Ratio Aferred Sto Ratio Amon Stock C C t of Capita	ock - % - % al - %	0 0 0 1 7 7	iation	30 0 30 0 0 0 0,5				
Total Cost - \$1000 Start of Evaluation	1309								1 1	Skalevis II	distail.	
Year		1985	1986	1987	1988	1989	1990	1991	1992	1993	1994	
Penaining Rook Value	0	0	0	0	0	0	0	0	1309	1265	1222	
Patura on Proferrad Stor	L I	0	0	0	0	0	0	0	0	0	0	
Potura on Common Stock		0	0	0	0	0	0	0	0	0	0	
Tetoroct		0	0	0	0	0	0	0	92	89	86	
Pook Depreciation		0	0	0	0	0	0	0	44	44	44	
Tax Depreciation		0	0	0	0	0	0	0	44	44	44	
Tax Depresident		0	0	0	0	0	0	0	0	0	0	
Deferred Taxes		0	0	0	0	0	0	0	0	0	0	
Descety Taxes		0	0	0	0	0	0	0	0	0	0	
Insurance		0	0	0	0	0	0	0	7	7	7	
Annual Carrying Charges		0	0	0	0	0	0	0	142	139	136	
Year		1995	1996	1997	1998	1999	2000	2001	2002	2003	2004	
Remaining Book Value		1178	1135	1091	1047	1004	960	916	873	829	785	
Return on Preferred Stor	k	0	0	0	0	0	0	0	0	0	0	
Return on Common Stock		0	0	0	0	0	0	0	0	0	0	
Interest		82	79	76	73	70	67	64	61	58	55	
Book Depreciation		44	44	44	44	44	44	44	44	44	44	
Tax Depreciation		44	44	44	44	44	44	44	44	44	44	
Income Taxes		0	0	0	0	0	0	0	0	0	0	
Deferred Taxes		0	0	0	0	0	0	0	0	0	0	
Property Tax		0	0	0	0	0	0	0	0	0	0	
Insurance		7	7	7	7	7	7	7	7	7	7	
Annual Carrying Charges		133	130	127	123	120	117	114	111	108	105	

ANNUAL	CARRYING	CHARGES,	PHASE	EIGHT	
				and the second second	

Investment for Phase	8		Economic Factors										
Initial Operation	1993	Pre	ferred St	ock Ratio		0		Book Life -	Yrs		30		
Cost Basis	1983	Reti	urn on Pr	eferred Sto	ock - %	0		Income Tax R	ate - %		0		
Capital Cost - \$1000		Com	mon Stock	Ratio		0		Tax Credit -	7.		0		
Heat Source	0	Ret	urn on Co	maon Stock	- %	0		Tax Life - Y	rs		30		
Piping	445	Deb	t Ratio			1		Accelerated	Tax Depres	iation	0		
Total	445	Deb	t Cost -	%		7		Property Tax	- %		0		
Escalation - \$1000	364	Wei	ghted Cos	t of Capita	al - %	7		Insurance Ra	te - %		0.5		
AFDC - \$1000	28												
Total Cost - \$1000	837												
Start of Evaluation	1985									Frederit	Tud?		
V		1005	1004	1007	1000	1000	1000	1001	1002	1007	1004		

Year	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
Remaining Book Value	0	0	0	0	0	0	0	0	837	809
Return on Preferred Stock	0	0	0	0	0	0	0	0	0	0
Return on Common Stock	0	0	0	0	0	0	0	0	0	0
Interest	0	0	0	0	0	0	0	0	59	57
Book Depreciation	0	0	0	0	0	0	0	0	28	28
Tax Depreciation	0	0	0	0	0	0	0	0	28	28
Income Taxes	0	0	0	0	0	0	0	0	0	0
Deferred Taxes	0	0	0	0	0	0	0	0	0	0
Property Tax	0	0	0	0	0	0	0	0	0	0
Insurance	0	0	0	0	0	0	0	0	4	4
Annual Carrying Charges	0	0	0	0	0	0	0	0	91	89
Year	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004
Remaining Book Value	781	754	726	698	670	642	614	586	558	530
Return on Preferred Stock	0	0	0	0	0	0	0	0	0	- 0
Return on Common Stock	0	0	0	0	0	0	0	0	0	0
Interest	55	53	51	49	47	45	43	41	39	37
Book Depreciation	28	28	28	28	28	28	28	28	28	28
Tax Depreciation	28	28	28	28	28	28	28	28	28	28
Income Taxes	0	0	0	0	0	0	0	0	0	0
Deferred Taxes	0	0	0	0	0	0	0	0	0	0
Property Tax	0	0	0	0	0	0	0	0	0	Û
Insurance	4	4	4	4	4	4	4	4	4	4
Annual Carrying Charges	87	85	83	81	79	77	75	73	71	69

## COMPOSITE CARRYING CHARGES, PHASES ONE THROUGH EIGHT

Phase	Year	Investment									Ecor	nomic	Facto	ors		1999-20		
	1005	4543		1	Prefer	red S	tock	Ratio	0		0		1	Book I	Life	- Yrs		30
2	1987	4833		1.00	Return	n on P	refe	erred !	Stock	- %	0		1	Incom	e Tax	Rate - %		0
34	1988	2174		1	Cosso	n Stor	k Ra	tio			0			Tax Ci	redit	- %		0
4	1989	2239		licit	Return	n on C		on Stor	ck -	7.	0			Tax L	ife -	Yrs		20
5	1990	1723			Debt I	Ratio					1		1	Accel	erate	d Tax Depre	ciation	0
6A	1991	3077		3.3	Debt I	Cost -	- %				7		1	Prope	rty T	ax - %		0
7	1992	1309		3-1	Weigh	ted Co	ost (	of Cap	ital	- %	7.00			Insur	ance	Rate - %		0.05
8	1993	837																
3B	1988	3146																
6B	1991	3801																
Ye	ar		1	1985		1986		1987		1988	1989		1990	2.2	1991	1992	1993	1994
Possining Po	ok Valu		1991	543		6325		10940	1	5566	17211	-	18181	2	3882	24132	23778	22588
Patura an Pr	afarrad	Stork		0		0		0		34	33		30		69	64	59	53
Return on Cr	anon St	ock		0		0		0		193	183		167		386	360	329	299
Totaract	and of ot	UCK		458		443		766		1072	1188		1257		1636	1656	1634	1553
Rook Denreri	ation			218		218		379		546	621		678		895	938	966	966
Tay Deprecia	ation			218		218		379		661	918		947		1275	1511	1477	1415
Income Taylor				0		0		0		136	43		40		203	89	86	85
Daforrad Tay	/BE			0		0		0		48	132		119		164	253	224	196
Property Tay				0		0		0		126	129		132		288	295	302	310
Insurance	that I a			33		33		57		83	95		103		138	144	148	148
Annual Carry	ying Cha	rges		709		694		1202		2238	2422		2527		3777	3799	3748	3611
																	2007	7004
Y	ear			1995		1996		1997		1998	1999	-	2000		2001	2002	2003	2004
Remaining B	ook Valu	16	2	1426	:	20292		19187		18110	17062		16042	712	15050	14087	.13152	12246
Return on P	referred	i Stock		48		44		40		36	32		29		26	24	22	20
Return on C	ommon St	tock		272		246		223		201	181		164		148	134	122	112
Interest				1475		1398		1322		1249	1178		1108		1040	974	909	847
Book Deprec	iation			966		966		966		966	966		966		966	966	966	966
Tax Depreci	ation			1353		1291		1230		1168	1106		1044		983	921	859	825
Income Taxe	5			86		89		93		100	109		119		132	146	162	168
Deferred Ta	xes			167		139		111		82	54		25		-3	-31	-60	-75
Property Ta	X			317		325		333		342	350		359		368	377	387	396
Insurance				148		148		148		148	148		148		148	148	148	148
Annual Carr	ying Ch	arges		3480	291	3355		3237		3125	3018		2919		2825	2738	2656	2582
												972						
																	Leii 10"3200	

## Table 9-12 (Sheet 1 of 2)

# CALCULATED UNIT COST FOR HOT WATER DISTRICT HEAT, 1985-1994

Start of Evaluation	1985					Econ	omic Factor	s			
Unit Costs	1983	Escalation	1	Preferred S	tock Ratio		0	E	ook Life -	Yrs	30
				Return on Pr	referred S	tock - %	0	I	ncome Tax	Rate - Z	0
Electricity - \$/NWh 5	51.44	7.5		Common Stoc	k Ratio		0	1	ax Credit	- 7	0
Pumping Power-\$/MWh 5	51.44	7.5		Return on Co	ommon Stoc	k - %	0	I	ax Life -	Yrs	30
Refuse Heat-\$/NWt	4.81	7.5		Debt Ratio			1	A	ccel. Tax	Deprec.	0
6as - \$/MBtu	5.13	7.5		Debt Cost -	7		7	1	nsurance R	ate - %	0.5
O&M Labor -\$/ManYr 3	5000	6.5		Weighted Co	st of Capi	tal - %	7	6	ross Recei	pts Tax-1	4
O&M Materials -% Cost		6.5									
Property Taxes - %	0	2.5									
Year		1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
A. Annual Quantities											
1. Heat Sales - MWt/yr		76490	76490	132762	164508	202458	211897	290804	307330	322157	322157
2. Refuse Heat - MWt/yr		0	0	100706	107139	111981	112850	117924	118895	119762	119762
3. Gas Consumed - MBtu/	/yr	350954	350954	169082	20131	80481	102959	82439	108548	141238	141238
4. Electricity Loss - 1	Wh/vr	0	0	0	11877	15419	16203	31625	33535	35128	35128
5. Pusping Energy - Mat	/vr	664	664	1152	1428	757	1839	2524	2667	2796	2796
6. D&M Labor - ManYr/vr		6	A	8	8	8	8	8	8	8	8
7. Oth Material-7 year	's Inv	0.457	0.457	1.007	1.617	1.857	1.007	1.00%	1.007	0.007	0.007
B. Unit Costs	5 101	01104	0.107			11004					
1. Replacement Elec	\$/MWh	59.45	63.90	68.70	75.14	80.78	86.83	98.34	105.72	113.65	122.17
2. Pumping Electricity-	-\$/MWh	59.45	63.90	68.70	73.85	79.39	85.34	91.74	98.62	106.02	113.97
3. Refuse Heat - \$/MWt		5.56	5.98	6.35	6.72	7.69	8.06	8.91	9.89	10.62	11.60
4. Fuel - \$/MBtu		5.93	6.37	6.85	7.36	7.92	8.51	9.15	9.84	10.57	11.37
5. D&M Labor - \$1000/Ma	anYr	40	42	45	48	51	54	58	62	66	70
C. Investments - \$1000 D. Annual Carrying Charge	es-\$100	6543 0/vr	0	4833	2174	2239	1723	3077	1309	837	0
1. Return on Preferred	Stock	0	0	0	34	33	30	69	64	59	53
2. Return on Common Stu	ock	0	0	0	193	183	167	386	360	329	299
3. Interest		458	443	766	1072	1188	1257	1636	1656	1634	1553
4. Book Depreciation		218	218	379	546	621	678	895	938	966	966
5 Tay Depreciation		218	218	379	661	918	947	1275	1511	1477	1415
6 Income Taylos		0	0	0	136	43	40	203	89	86	85
7 Deferred Taxes		0	0	0	48	132	119	164	253	224	196
O Property Tax		0	0	0	126	179	132	286	295	302	310
O. Frupercy tax		77	77	57	87	05	107	138	144	148	148
7. Insurance											
Sub- Total		709	694	1202	2238	2422	2527	3777	3799	3748	3611
E. Operating Expenses -	\$1000/y	r									
1. Replacement Electri	city	0	0	0	892	1245	1407	3110	3545	3992	4292
2. Refuse Heat		0	0	639	720	861	910	1051	1176	1272	1389
3. Fuel		2081	2237	1158	148	637	876	754	1068	1493	1605
4. O&M Labor		238	254	360	384	409	435	463	494	526	560
5. O&M Materials		29	31	81	122	171	200	243	272	290	309
6. Pumping Cost		39	42	79	105	60	157	232	263	296	319
Sub- Total		2387	2564	2319	2372	3384	3984	5853	6817	7869	8473
F. Gross Receipts Tax-\$1	000/yr	124	130	141	184	232	260	385	425	465	483
6. Required Revenues- \$1	000/yr	3220	3388	3661	4794	6038	6772	10016	11041	12082	12548
H Unit Cost of Host - 4	/MRt										
1 Fived Evenness	- III LU	\$7.72	\$2.66	\$2.45	99 52	\$7.51	\$7.49	\$3.81	\$3.67	\$7.45	\$7.28
2 Replacement Flortei	rity	\$0.00	\$0.00	10.00	\$1.50	\$1.80	\$1.95	57.17	\$7.78	51 52	190.72
3. Anerating Evapores		\$9.15	\$9.97	\$5.12	\$7.63	\$0.72	\$3.56	\$2.76	\$3.12	\$7.57	\$7.80
4. Gross Receipte Tay		\$0.47	\$0.50	\$0.31	57.02	\$0.34	\$0.36	\$0.39	\$0.40	\$0.42	\$0.44
a arass necespes lax											
		\$12.33	\$12.98	\$8.08	\$8.54	\$8.74	\$9.36	\$10.09	\$10.53	\$10.99	\$11.43

### Table 9-12 (Sheet 2 of 2)

# CALCULATED UNIT COST FOR HOT WATER DISTRICT HEAT, 1995-2004

Unit Costs    1985    Esculation    Preferred Stoct 7 L    0    Bub Life - Yrs    30      Electricity - 4//Mh    31.44    7.5    Cason Stock Fails    0    Trave Ta Fairo    0    Trave    0    Trave Ta Fairo    Trave Ta Fairo	Start of Evaluation 1985 Economic Factors											
Clearticity - <i>MYMA</i> S1.44    7.5    Cosson Stock Fills    0    Income Tax Hile - 1    0      Pusping Power- <i>MYMA</i> S1.44    7.5    Return on Cosson Stock - 1    0    Tax Life - 1'rs    30      Bruise Nucl.*/MYMA    S1.14    7.5    Beet Ratio    0    Tax Life - 1'rs    30      Bruise Nucl.*/MYMA    S1.13    7.5    Beet Cost - 1    7    Tax Life - 1'rs    30      Bruise Nucl.*/MYMA    S1.13    7.5    Beet Cost - 1    7    Froms Receipts Tax - 1    4      BR Labor - HYLAN    S200    2001    2002    2002    2003    2004      A. Annual Datatities    Intext Sales - NUL/Y    S2157    S22157    S2215	Unit Costs	1983	Escalation	P	referred St	ock Ratio		0	Bo	ook Life -	Yrs	30
Electricity - 4/AMM 51.44 7.5 Common Stock Tailing 0 Tar Credit - 3 0 Tar Life - 1's 30 Arcel. Tar. Hopere. 0 Tar Life - 1's 30 Arcel. Tar. Hopere. 0 Tar Life - 1's 30 Arcel. Tar. Hopere. 0 Tar Life - 1's 30 Arcel. Tar. Hopere. 0 Tar Life - 1's 30 Arcel. Tar. Hopere. 0 Tar Life - 1's 30 Arcel. Tar. Hopere. 0 Tar 1's 5000 6.5 Keighted Cost of Capital - 1 7 Tar. Tar. Hopere. 0 Tar 1's 5000 6.5 Keighted Cost of Capital - 1 7 Tar 1's 5000 7 Tar 1's 50000 7 Tar 1's 5000 7 Tar 1's 50000 7 Tar 1's				R	eturn on Pr	eferred St	ock - %	0	In	ncome Tax F	Rate - Z	0
Dupping Power-f/NM    S1.44    7.5    Beth Ratio    Beth Ratio    Marcell - Tail    Marcell - Tail <th< td=""><td>Electricity - \$/MWh 5</td><td>1.44</td><td>7.5</td><td>2000</td><td>ionaon Stock</td><td>Ratio</td><td></td><td>0</td><td>Ta</td><td>ax Credit -</td><td>- 7</td><td>0</td></th<>	Electricity - \$/MWh 5	1.44	7.5	2000	ionaon Stock	Ratio		0	Ta	ax Credit -	- 7	0
Refuse Real-F/MRI 4.81 7.5 Debt Relio 1 McCell 12819782. 00 Ses - F/MBL 5.13 7.5 Debt Cost -1 T Fiberance Rate -1 0.5 Beross Receipts Tax-2 4 Wear 195 1996 1997 1998 1999 2000 2001 2002 2003 2004 A. Annual Quantities A. Annual Quantities B. B. B	Pusping Power-\$/KWh 5	1.44	7.5	R	teturn on Co	ason Stock	( - X	0	Ti	ax Life -	rs	20
Bas - AFRE    S.13    7.5    Debt Cost - 1    //    Insurance Acte - 1    0      DRM Noterials - 1 Cost	Refuse Heat-\$/MWt	4.81	7.5	I	bt Ratio			1	R	cel. lax I	veprec.	0.5
DBM Labor - #/Man'r  35000  6.5  Weighted Cost of Capital - 1  /  Brogs Methys int - 4    Waar  1995  1996  1997  1998  1999  2000  2001  2002  2003  2204    A. Annual Bunktities  1.  1.  1995  119762  <	6as - \$/MBtu	5.13	7.5	1	lebt Cost -	1		1	11	nsurance Ka		0.5
DAR Materials -2 Losi	D&M Labor -\$/ManYr 3	35000	6.5		leighted Cos	t of Capit	tal - L	1	61	ross Recei	DES Idx-4	
Property Taxes - 1    0      Year    1995    1996    1997    1998    1999    2000    2001    2002    2003    2204      A. Annual Guntities    . <t< td=""><td>O&amp;M Materials -% Cost</td><td></td><td>6.5</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	O&M Materials -% Cost		6.5									
Year    1995    1996    1997    1998    1999    2000    2001    2002    2003    2004      A. Annual Guntities    1.    Net Sales - MK/yr    322157    3221754    2776    2776	Property Taxes - 1	0										
A. Annual Guantities 1. Nest Sales - NML/yr 2. Refuse Matter - MWL/yr 3. Gas Consumed - MBUL/yr 4. Electricity Loss - MML/yr 4. Electricity Loss - MML/yr 5. Dift Loss - MML/yr 4. Electricity Loss - MML/yr 5. Dift Loss - Sill S. Sill Sill	Year		1995	1996	1997	1998	1999	2000	2001	2002	2003	2004
1. Heat Sales - MML/yr 22157 32152 35128 3518 35128 35128 35128 35128 35128 3518 35128 35128 35128 35128 35128 3518 35128 35128 3518 35128 35128 35128 3518 35128 35128 3518 35128 35128 3518 35128 35128 3518 35128 3518 35128 3518 3518 35128 3518 3518 35128 3518 3518 3518 3518 35128 3518 3518 3518 3518 3518 3518 3518 351	A. Annual Quantities					120	BOINT N				700157	700157
2. Refuse Heat - MH/yr 119762 11976 1108 1040 9774 909 115 113 11164 1164 1164 1134 1122 1122 1123 11164 1164 1164 1134 1122 1122 1123 1164 1164 1164 1134 1122 1122 113. 1164 1164 1164 1134 1122 1122 113. 1164 1164 1164 1134 1122 1122 113. 1164 1164 1164 1134 1122 1122 113. 1164 1164 1164 1134 1122 1122 113. 1164 1164 1164 1164 1164 1164 1162 1162	1. Heat Sales - MWt/yr		322157	322157	322157	322157	322157	322157	322157	322137	522157	322137
3. Eas Consumed - HRU/yr 141238 14128 1418 14188 14148 14148 14148 14148 14148	2. Refuse Heat - MWt/yr	r	119762	119762	119762	119762	119762	119762	119762	119/62	119/62	117/02
4. Electricity Loss - MMA/Yr  35120  <	3. Gas Consumed - MBtu.	/yr	141238	141238	141238	141238	141238	141238	141238	141238	141238	191230
5. Penging Energy - Nah/yr 2796 2796 2796 2796 2796 2796 2796 2796	4. Electricity Loss - I	MWh/yr	35128	35128	35128	35128	35128	35128	35128	30128	33128	33120
6. OLM Labor - ManTr/yr 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	5. Pumping Energy - Mw	h/yr	2796	2796	2796	2796	2796	2798	2/96	2/96	2170	2170
7. QLM Material-X year's Iav 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.0000 0.0000 0.000 0.000 0.000 0.000 0.000 0.000 0.000	6. O&M Labor - ManYr/y	r	8	8	8	8	8	8	8	0 007	0 007	0 003
D. Unit Costs 1. Replacement Elec 4/NWh 131.33 141.16 151.77 163.15 175.39 188.54 202.68 217.89 234.23 251.79 2. Pupping Electricity-4/NWh 122.32 131.71 141.59 152.20 163.642 175.89 189.08 203.26 216.51 234.90 3. Refuse Heat - 5/NWt 12.23 131.55 151.4 17.09 18.68 21.25 22.47 25.46 26.62 26.13 4. Fuel - 4/YBtu 12.22 13.13 14.12 15.18 16.32 17.54 18.66 20.27 21.79 23.43 5. UMH Labor - 51000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7. O&M Material-% year	's Inv	0.00%	0.00%	0.00%	0.00%	0.001	0.001	0.001	0.001	0.00%	0.001
1. heplatemin tits J. Man. 10.25 1.141.59 152.20 163.62 175.89 199.08 203.26 218.51 234.90 2. Pusping Electricity - J.Man. 12.23 131.71 141.59 152.20 163.62 17.54 18.08 21.25 22.47 22.47 22.47 23.43 4. Fuel - \$1000/ManYr 75 79 85 90 96 102 109 116 123 131 C. Investments - \$1000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	B. Unit Costs	\$ / NMb	171 77	141.18	151.77	163.15	175.39	188.54	202.68	217.89	234.23	251.79
2. Pedging Electricity Film  12.22  13.35  15.14  17.09  18.68  21.25  22.47  25.46  26.62  26.13    13.85  OMA Labor - \$1000/ManYr  75  79  85  90  96  102  109  116  123  131    C. Investents - \$1000  <	1. Replacement clec	*/ (IWI)	122 52	131 71	141.59	152.20	163.62	175.89	189.08	203.26	218.51	234.90
3. Artuse neat - 3/Int.  11.13.5 <t< td=""><td>Z. Pumping Electricity</td><td>-4/11811</td><td>12 33</td><td>17.55</td><td>15.14</td><td>17.09</td><td>18.68</td><td>21.25</td><td>22.47</td><td>25.46</td><td>26.62</td><td>26.13</td></t<>	Z. Pumping Electricity	-4/11811	12 33	17.55	15.14	17.09	18.68	21.25	22.47	25.46	26.62	26.13
4. Public */Table  12.12  12.13  12.14  12.13  131    5. OHA Labor - \$1000/ManYr  75  79  85  90  96  102  109  116  123  131    C. Investents - \$1000  0	S. Ketuse Heat - P/INL		12.33	13.00	14 12	15.18	16.32	17.54	18.86	20.27	21.79	23.43
C. Investments - \$1000  0 <td>5. OWN Labor - \$1000/M</td> <td>an¥r</td> <td>75</td> <td>79</td> <td>85</td> <td>90</td> <td>96</td> <td>102</td> <td>109</td> <td>116</td> <td>123</td> <td>131</td>	5. OWN Labor - \$1000/M	an¥r	75	79	85	90	96	102	109	116	123	131
D. Annual Carrying Charges-5100 1. Return on Preferred Stock 48 44 40 36 32 29 26 24 22 20 2. Return on Chamaon Stock 272 246 223 201 181 164 148 134 122 112 3. Interest 1475 1378 1322 1249 1178 1040 974 909 847 4. Book Depreciation 1533 1291 1230 1168 1106 1044 983 921 859 823 6. Income Taxes 86 89 93 100 109 119 132 146 162 166 7. Deferred Taxes 167 139 111 82 54 25 -3 -31 -60 -75 8. Property Tax 317 325 333 342 350 359 368 377 387 396 9. Insurance 148 148 148 148 148 148 148 148 148 148	C. Investments - \$1000		0	0	0	0	0	0	0	0	0	0
1. Return on Preferred Stock 48 44 40 35 32 22 24 23 201 181 164 168 134 122 112 2. Return on Common Stock 272 246 223 201 181 164 168 134 122 112 3. Interest 1475 1398 1322 1249 1178 1108 1040 974 909 547 4. Book Depreciation 966 966 966 966 966 966 966 966 966 96	D. Annual Carrying Charg	es-\$10	0		10	71	79	20	26	74	22	20
2. Return on Losson Stock 272 245 223 247 1178 1108 1040 774 709 B47 3. Interest 1475 1398 1322 1249 1178 1108 1040 774 709 B47 4. Book Depreciation 966 966 966 966 966 966 966 966 966 96	1. Return on Preferred	Stock	48	99	40	201	101	164	148	134	172	112
3. Interest  1475  1376  1322  1247  1176  105  106  767  73  73  73  73  7367  3767	2. Return on Common St	OCK	212	1700	1722	1240	1170	1108	1040	974	909	E47
4. Book Depreciation  766  767  767  767  767  767  737  367  377  387  376    9. Insurance  148	3. Interest		14/5	1248	1322	011	0111	1100	946	966	966	966
5. Tax Depreciation 1333 1271 1230 1188 1108 109 109 119 132 146 152 146 6. Income Taxes 86 89 93 100 109 119 132 146 152 146 7. Deferred Taxes 167 139 111 82 54 25 -3 -31 -60 -75 8. Property Tax 317 325 333 342 350 359 368 377 387 376 9. Insurance 148 148 148 148 148 148 148 148 148 148	4. Book Depreciation		966	966	100	1140	1104	1044	700	921	859	875
6.  Income Taxes  80  67  73  100  101  111  102  110  102  111  111  111  1111  1111  1111  1111  1111  1111	5. Tax Depreciation		1353	1291	1230	1100	100	110	132	146	162	168
7. Deferred laxes  16/  137  111  325  313  342  350  359  368  377  387  376    8. Property Tax  317  325  333  342  350  359  368  377  387  376    9. Insurance  148<	6. Income Taxes		80	170	75	100	54	25	7-	-31	-60	-75
B. Property Tax  317  325  333  342  330  307  308  309	7. Deferred laxes		16/	137	777	742	750	150	847	377	387	396
9. Insurance  148	B. Property Tax		517	323	333	140	140	148	148	148	148	148
Sub- Total  3480  3355  3237  3125  3018  2919  2825  2738  2656  2582    E. Operating Expenses - \$1000/ 1. Replacement Electricity  4613  4959  5331  5731  6161  6623  7120  7654  8228  8945    2. Refuse Heat  1477  1623  1813  2047  2237  2545  2691  3048  3129    3. Fuel  1726  1855  1994  2144  2305  2477  2663  2843  3078  3305    4. D&M Labor  596  635  676  720  767  817  870  926  987  1251    5. DEM Materials  329  350  373  397  423  451  480  511  544  586    6. Pumping Cost  343  368  376  426  457  492  529  568  611  657    5ub- Total  9083  9791  10584  11465  12350  13405  14353  15572  16636  17576    6. Required Revenues- \$1000/yr	9. Insurance		148	148	140	140						
E. Operating Expenses - \$1000/ 1. Replacement Electricity 4613 4959 5331 5731 6161 6623 7120 7654 8228 8945 2. Refuse Heat 1477 1623 1813 2047 2237 2545 2691 3049 3188 3129 3. Fuel 1726 1855 1994 2144 2305 2477 2663 2863 3078 3309 4. OEM Labor 596 635 676 720 767 817 870 926 987 1051 5. OEM Materials 329 350 373 397 423 451 480 511 544 586 6. Pumping Cost 343 368 396 426 457 492 529 568 611 657 Sub- Total 9083 9791 10584 11465 12350 13405 14353 15572 16636 17570 F. Gross Receipts Tax-\$1000/yr 503 526 553 584 615 653 687 732 772 800 G. Required Revenues- \$1000/yr 13066 13672 14374 15173 15983 16976 17865 19042 20064 20950 H. Unit Cost of Heat - \$/MBtu 1. Fixed Expenses \$3.17 \$3.05 \$2.94 \$2.84 \$2.75 \$2.65 \$2.57 \$2.49 \$2.42 \$2.33 2. Replacement Electricity \$4.20 \$4.51 \$4.85 \$5.21 \$5.60 \$6.02 \$6.48 \$6.96 \$7.48 \$2.00 3. Operating Expenses \$4.07 \$4.39 \$4.78 \$5.21 \$5.63 \$6.17 \$6.58 \$7.20 \$7.65 \$7.9 4. Gross Receipts Tax \$0.46 \$0.48 \$0.50 \$0.53 \$0.56 \$0.59 \$0.62 \$0.67 \$0.70 \$2.7	Sub- Total		3480	3355	3237	3125	3018	2919	2825	273B	2656	2582
1. Replacement Electricity  4613  4739  5331  5741  5047  52477  2663  2863  3078  3330    3. Fuel  1726  1855  1994  2144  2305  2477  2663  2863  3078  3309    4. D&H Labor  596  635  676  720  767  817  870  926  987  1251    5. D&M Materials  329  350  373  397  423  451  480  511  544  580    6. Pumping Cost  343  368  376  426  457  492  529  568  611  657    Sub- Total  9083  9791  10584  11465  12350  13405  14353  15572  16636  17570    F. Gross Receipts Tax-\$1000/yr  13066  13672  14374  15173	E. Operating Expenses -	\$1000/				E 774	1111	11.77	7120	7454	8228	RRAS
2. Refuse Heat  14/7  1623  1813  2047  2237  2043  2041  5071<	1. Replacement Electri	icity	4613	4939	3331	3/31	1010	2545	2601	7019	3188	3129
3. Fuel  1726  1855  1994  2144  2305  2477  2665  3655  576    4. D&M Labor  596  635  676  720  767  B17  870  926  987  1051    5. D&M Materials  329  350  373  397  423  451  480  511  544  586    6. Pumping Cost  343  368  396  426  457  492  529  568  611  657    Sub- Total  9083  9791  10584  11465  12350  13405  14353  15572  16636  17570    F. Gross Receipts Tax-\$1000/yr  503  526  553  584  615  653  687  732  772  800    6. Required Revenues- \$1000/yr  13066  13672  14374  15173  15983  16976  17865  19042  20064  20950    H. Unit Cost of Heat - \$/MBtu  1. Fixed Expenses  \$3.17  \$3.05  \$2.94  \$2.84  \$2.75  \$2.65  \$2.57  \$2.49  \$2.42  \$2.33	2. Refuse Heat		1477	1623	1813	204/	2231	2343	2071	2017	3078	1110
4. D&H Labor  596  635  676  720  767  617	3. Fuel		1726	1822	1994	2144	2303	29/1	2003	074	007	1051
5. DEM Materials  329  330  373  377  423  431  400  311  544  557    6. Pumping Cost  343  368  376  426  457  492  529  568  611  657    Sub- Total  9083  9791  10584  11465  12350  13405  14353  15572  16636  17570    F. Gross Receipts Tax-\$1000/yr  503  526  553  584  615  653  687  732  772  800    6. Required Revenues- \$1000/yr  13066  13672  14374  15173  15983  16976  17865  19042  20064  20950    H. Unit Cost of Heat - \$/MBtu  1. Fixed Expenses  \$3.17  \$3.05  \$2.94  \$2.84  \$2.75  \$2.65  \$2.57  \$2.49  \$2.42  \$2.33    2. Replacement Electricity  \$4.20  \$4.51  \$4.65  \$5.21  \$5.63  \$6.17  \$6.58  \$7.20  \$7.65  \$7.9    4. Gross Receipts Tax  \$0.46  \$0.48  \$0.50  \$0.53  \$0.56  \$0.59  \$0.62	4. D&M Labor		596	635	6/6	720	/0/	451	400	511	544	580
6. Pumping Cost  343  368  346  426  437  442  327  368  011	5. O&M Materials		329	350	5/5	397	423	402	90V 520	548	611	657
Sub- Total  9083  9791  10584  11465  12350  13405  14353  15572  16636  17570    F. Gross Receipts Tax-\$1000/yr  503  526  553  584  615  653  687  732  772  804    G. Required Revenues- \$1000/yr  13066  13672  14374  15173  15983  16976  17865  19042  20064  20956    H. Unit Cost of Heat - \$/MBtu  \$3.17  \$3.05  \$2.94  \$2.84  \$2.75  \$2.65  \$2.57  \$2.49  \$2.42  \$2.33    2. Replacement Electricity  \$4.20  \$4.51  \$4.65  \$5.21  \$5.63  \$6.17  \$6.58  \$7.20  \$7.48  \$2.03    3. Operating Expenses  \$4.07  \$4.39  \$4.78  \$5.21  \$5.63  \$6.17  \$6.58  \$7.20  \$7.65  \$7.99    4. Gross Receipts Tax  \$0.46  \$0.48  \$0.50  \$0.53  \$0.56  \$0.59  \$0.62  \$0.67  \$0.70  \$7.77	6. Pumping Cost		343		249	420	104	472				
F. Gross Receipts Tax-\$1000/yr  503  526  553  584  615  653  687  732  772  804    G. Required Revenues- \$1000/yr  13066  13672  14374  15173  15983  16976  17865  19042  20064  20956    H. Unit Cost of Heat - \$/NBtu	Sub- Total		9083	9791	10584	11465	12350	13405	14353	15572	16636	17570
6. Required Revenues- \$1000/yr  13066  13672  14374  15173  15983  16976  17865  19042  20064  20956    H. Unit Cost of Heat - \$/MBtu  1. Fixed Expenses  \$3.17  \$3.05  \$2.94  \$2.84  \$2.75  \$2.65  \$2.57  \$2.49  \$2.42  \$2.33    2. Replacement Electricity  \$4.20  \$4.51  \$4.85  \$5.21  \$5.60  \$6.02  \$6.48  \$6.96  \$7.48  \$8.00    3. Dperating Expenses  \$4.07  \$4.39  \$4.78  \$5.21  \$5.63  \$6.17  \$6.58  \$7.20  \$7.65  \$7.9    4. Gross Receipts Tax  \$0.46  \$0.48  \$0.50  \$0.53  \$0.56  \$0.57  \$0.62  \$0.67  \$0.70  \$2.77	F. Gross Receipts Tax-\$	1000/yr	503	526	553	584	615	653	687	732	772	80
H. Unit Cost of Heat - \$/MBtu    1. Fixed Expenses  \$3.17  \$3.05  \$2.94  \$2.84  \$2.75  \$2.65  \$2.57  \$2.49  \$2.42  \$2.32    2. Replacement Electricity  \$4.20  \$4.51  \$4.85  \$5.21  \$5.60  \$6.02  \$6.48  \$6.96  \$7.48  \$8.00    3. Dperating Expenses  \$4.07  \$4.39  \$4.78  \$5.21  \$5.63  \$6.17  \$6.58  \$7.20  \$7.65  \$7.9    4. Gross Receipts Tax  \$0.46  \$0.48  \$0.50  \$0.53  \$0.56  \$0.59  \$0.62  \$0.67  \$0.70  \$0.7	6. Required Revenues- \$	1000/y	13066	13672	14374	15173	15983	16976	17865	19042	20064	2095
1. Fixed Expenses  \$3.17  \$3.05  \$2.94  \$2.84  \$2.75  \$2.65  \$2.57  \$2.49  \$2.42  \$2.33    2. Replacement Electricity  \$4.20  \$4.51  \$4.65  \$5.21  \$5.60  \$6.02  \$6.48  \$6.96  \$7.48  \$2.03    3. Operating Expenses  \$4.07  \$4.39  \$4.78  \$5.21  \$5.63  \$6.17  \$6.58  \$7.20  \$7.65  \$7.9    4. Gross Receipts Tax  \$0.46  \$0.48  \$0.50  \$0.53  \$0.56  \$0.59  \$0.62  \$0.67  \$0.70  \$2.7	H. Unit Cost of Heat -	\$/MBtu	1					5.000				
2. Replacement Electricity  \$4.20  \$4.51  \$4.85  \$5.21  \$5.60  \$6.02  \$6.48  \$6.96  \$7.48  \$2.0    3. Operating Expenses  \$4.07  \$4.39  \$4.78  \$5.21  \$5.63  \$6.17  \$6.58  \$7.20  \$7.65  \$7.9    4. Gross Receipts Tax  \$0.46  \$0.48  \$0.50  \$0.53  \$0.56  \$0.59  \$0.62  \$0.67  \$0.70  \$2.7	1. Fixed Expenses		\$3.17	\$3.05	\$2.94	\$2.84	\$2.75	\$2.65	\$2.57	\$2.49	\$2.42	\$2.3
3. Operating Expenses    \$4.07    \$4.39    \$4.78    \$5.21    \$5.63    \$6.17    \$6.58    \$7.20    \$7.65    \$7.9      4. Gross Receipts Tax    \$0.46    \$0.48    \$0.50    \$0.53    \$0.56    \$0.59    \$0.62    \$0.67    \$0.70    \$2.7	2. Replacement Electr	icity	\$4.20	\$4.51	\$4.85	\$5.21	\$5.60	\$6.02	\$6.48	\$6.96	\$7.48	\$8.0
4. Gross Receipts Tax \$0.46 \$0.48 \$0.50 \$0.53 \$0.56 \$0.59 \$0.62 \$0.67 \$0.70 \$2.7	3. Operating Expenses	5	\$4.07	\$4.39	\$4.78	\$5.21	\$5.63	\$6.17	\$6.58	\$7.20	\$7.65	\$1.9
	4. Gross Receipts Tax	1	\$0.46	\$0.48	\$0.50	\$0.53	\$0.56	\$0.59	\$0.62	\$0.67	\$0.70	\$2.7
\$11.88 \$12.43 \$13.07 \$13.80 \$14.54 \$15.44 \$16.25 \$17.52 \$18.20 \$19.0			\$11.88	\$12.43	\$13.07	\$13.80	\$14.54	\$15.44	\$16.25	\$17.32	\$18.25	\$19.0

### HOT WATER RADIATION CONSUMER RETROFIT PAYBACK

			CO	ST COMPARI	SON FOR:	HOT WATER 1984					
				HOOK	UP YEAR:	1985					
~		TOTAL CONVI	ERSION CO	ST (PRESEN	T YEAR):	\$55,000					
		10.00		ESCALATI	ON RATE:	6.50%				the line ask	
		TOTAL CON	VERSION C	OST (HOOKU	P YEAR):	\$58,575	(	Cost:	\$55/kW	t	
				PERCENT F	INANCED:	100%					
				CASH INV	ESTMENT:	\$0					
			TER	M OF LOAN	(YEARS):	20					
				INTERE	ST RATE:	12.00%					
				ANNUAL	PAYMENT:	\$7,842					
				T	AX RATE:	50%					
			EX	PENSING DE	DUCTION:	\$5,000					
		102	YEAR	S OF DEPRE	CIATION:	5					
	AC	CELERATED	DEPRECIAT	ION? (1=YE	5,0=N0):	1					
PC	ITENTIAL EN	D USE ENER	GY SAVING	S FROM CON	VERSION:	20					
		ESTIMATED	CURRENT	SYSTEM EFF	ICIENCY:	551					
			HEATI	NG EQUIPME	NT TYPE:	OIL BOILER	1				
							CURREN	IT ANNUAL	FUEL USE -		
					S		PRIMARY	BACKUP	BACKUP	TOTAL	
			1196	FU	EL TYPE:	(	DIL			TYL TO THE	
			CONSUMPT	ION (MILLI	ON BTU):		13843			13843	
	ES	STIMATED DH	CONSUMPT	ION (MILLI	ON BTU):					7614	
		CURRENT	FUEL RAT	E (\$/MILLI	ON BTUI:	8.43					
			FUE	L ESCALATI	ON RATE:	7.50%					
	CUPPENT	DH		1	DISTRICT	HEATING -	1		DISTRICT		ENERGY
	FIIFI	FIIFI	CHIPPENT		CUNNE	RETING		TAY	HEATING		COST
	PATE	PATE	ENEDCA	ENERGY	AMORTI	7ATION	TOTAL	FEFERTS	SAVINES	SAUTNES	SAUTNES
YEAR	\$/MMBTU	\$/MMBTU	COSTS	COSTS	PRINC.	INTER.	COSTS	(\$)	(\$)	(\$)	(\$)
									(01		Larged Lines
1985	9.06	12.33	125449	93876	813	7029	101718	-5754	17977	17977	31572
1986	9.74	12.98	134857	98825	911	6931	106667	-8657	19533	37510	36032
1987	10.47	8.08	144972	61518	1020	6822	69360	-32690	42921	80431	83453
1988	11.26	8.54	155845	65021	1142	6700	72863	-36437	46545	126977	90824
1989	12.10	8.74	167533	66543	1279	6563	74385	-41588	51560	178536	100990
1990	13.01	9.36	180098	71264	1433	6409	79106	-51212	49780	228316	108834
1991	13.99	10.09	193605	76822	1605	6237	84664	-55273	53668	281984	116783
1992	15.03	10.53	208126	80172	1797	6045	88014	-60955	59157	341142	127954
1993	16.16	10.99	223735	83674	2013	5829	91516	-67116	65103	406245	140061
1994	17.37	11.43	240515	87024	2254	5588	94866	-73952	71697	477942	153491
1995	18.68	11.88	258554	90450	2525	5317	98292	-81393	78868	556811	168104
1996	20.08	12.43	277945	94638	2828	5014	102480	-89147	86319	643130	183308
1997	21.58	13.07	298791	99510	3167	4675	107352	-97303	94136	737265	199281
1998	23.20	13.80	321201	105068	3547	4295	112910	-105919	102371	839637	216132
1999	24.94	14.54	345291	110702	3973	3869	118544	-115360	111387	951023	234588
2000	26.81	15.44	371187	117555	4450	3392	125397	-125120	120670	1071694	253633
2001	28.83	16.25	399026	123722	4984	2858	131564	-136223	131239	1202933	275305
2002	30.99	17.32	428953	131868	5582	2260	139710	-147412	141831	1344764	297085
2003	33.31	18.25	461125	138949	6252	1590	146791	-160293	154041	1498805	322176
2004	35.81	19.06	495709	145116	7002	840	152958	-174876	167875	1666680	350593

PAYBACK (NO FINANCING) IN YEAR: 1987

PAYBACK (WITH FINANCING) IN YEAR: 1987

9-18

-	1 7	L	0	7 /	
1 7	b	0	· U -	_ 1/	
10		E	7-	- 1-	
		-	~	_	

COST COMPARISON FOR:	STEAM-2P					
PRESENT YEAR:	1984					
HOOKUP YEAR:	1985					
TOTAL CONVERSION COST (PRESENT YEAR):	\$176,000					
ESCALATION RATE:	6.50%					
TOTAL CONVERSION COST (HOOKUP YEAR):	\$187,440					
PERCENT FINANCED:	100%					
CASH INVESTMENT:	\$0	<b>C</b>	±. ¢1	75/14	-	
TERM OF LOAN (YEARS):	20	Cos	τ: 31	L/5/KWI	L	
INTEREST RATE:	12.00%					
ANNUAL PAYMENT	\$25,094					
TAX RATE:	: 50%					
EXPENSING DEDUCTION	\$5,000					
YEARS OF DEPRECIATION	: 5					
ACCELERATED DEPRECIATION? (1=YES,0=ND)	1 1					
POTENTIAL END USE ENERGY SAVINGS FROM CONVERSION	: 13%					
ESTIMATED CURRENT SYSTEM EFFICIENCY	: 55%					
HEATING EQUIPMENT TYPE	OIL BUILER	CURDENT				
		DDIMADY	BACKIID	PACYIIP	TOTAL	
		PRIMAT	DHUKUF	DHUNUT	TOTAL	
FUEL TIPE		17047			13843	
CONSUMPTION (AILLION BID)		13043			5814	
ESTIMATED DH CUNSUMPTION (HILLION BIO)	. 0.47				0011	
CURKENT FUEL KATE (\$/AILLIUN BID)	7 507					
FUEL ESCALATION RATE	1.004					
CURRENT DH I DISTRIC	T HEATING	I		DISTRICT		
FUEL FUEL CURRENT CONV	ERSION		TAX	HEATING	CUMULATIVE	
RATE RATE ENERGY ENERGY AMORT	TIZATION	TOTAL	EFFECTS	SAVINGS	SAVINGS	
AR \$/MMBTU \$/MMBTU COSTS COSTS PRINC.	INTER.	COSTS	(\$)	(\$)	(\$)	

.....

TWO PIPF	STEAM	RADIATION	CONSUMER	RETROFIT	PAYBACK
----------	-------	-----------	----------	----------	---------

ENERGY COST AVINGS (\$) YE 12.33 9.06 9.74 12.98 -18925 10.47 8.08 -23220 8.54 11.26 -28702 12.10 8.74 -52584 13.01 9.36 -57491 13.99 10.09 -63780 10.53 15.03 -70593 10.99 16.16 -78090 17.37 11.43 -86234 11.88 18.68 -94816 6970B3 20.08 12.43 -103921 13.07 21.58 -113612 23.20 13.80 14.54 -124187 24.94 -135282 26.81 15.44 -147701 16.25 28.83 -160511 30.99 17.32 -174965 18,25 33.31 -191102 35.81 19.06 

PAYBAC	K (NO	FINANCING)	IN	YEAR:	1988	
PAYEACK	(WITH	FINANCING)	IN	YEAR:	1968	

PAYEACK (WITH FINANCING) IN TEAK:

### ONE PIPE STEAM RADIATION CONSUMER RETROFIT PAYBACK

COST COMPARISON FOR: STE	EAM-IP			1	
PRESENT YEAR:	1984			1	
HOOKUP YEAR:	1985				
TOTAL CONVERSION COST (PRESENT YEAR):\$50	00,000			stats said	
ESCALATION RATE:	6.50%				
TOTAL CONVERSION COST (HOOKUP YEAR):\$53	32,500				
PERCENT FINANCED:	100%				
CASH INVESTMENT:	\$0	Cos	+ . \$5	00/1/1+	
TERM OF LOAN (YEARS):	20	003	ι. ψυ	OUT KWC	
INTEREST RATE:	12.00%				
ANNUAL PAYMENT: \$7	71,290				
TAX RATE:	50%				
EXPENSING DEDUCTION: \$	\$5,000				
YEARS OF DEPRECIATION:	5				
ACCELERATED DEPRECIATION? (1=YE5,0=ND):	1				
POTENTIAL END USE ENERGY SAVINGS FROM CONVERSION:	15%				
ESTIMATED CURRENT SYSTEM EFFICIENCY:	55%				
HEATING EQUIPMENT TYPE: DIL	BOILER				
		CURRENT	ANNUAL F	UEL USE	
		PRIMARY	BACKUP	BACKUP	TOTAL
FUEL TYPE:	OI	L			
CONSUMPTION (MILLION BTU):		13843			13843
ESTIMATED DH CONSUMPTION (MILLION BTU):					5537

CURRENT FUEL RATE (\$/MILLION BTU): 8.43 FUEL ESCALATION RATE: 7.50%

	CURRENT	DH		I	DISTRICT	HEATING	1		DISTRICT		ENERGY
	FUEL	FUEL	CURRENT		CONVE	RSION		TAX	HEATING	CUMULATIVE	COST
	RATE	RATE	ENERGY	ENERGY	AMORTI	ZATION	TOTAL	EFFECTS	SAVINGS	SAVINGS	SAVINGS
YEAR	\$/MMBTU	\$/MMBTU	COSTS	COSTS	PRINC.	INTER.	COSTS	(\$)	(\$)	(\$)	(\$)
1985	9.06	12.33	125449	68274	7390	63900	139564	45425	31310	31310	57175
1986	9.74	12.98	134857	71873	8277	63013	143163	58039	49733	81043	62985
1987	10.47	8.08	144972	44741	9271	62020	116031	36282	65223	146265	100231
1988	11.26	8.54	155845	47288	10383	60907	118578	31563	68829	215095	108557
1989	12.10	8.74	167533	48395	11629	59661	119696	25649	73497	288591	119138
1990	13.01	9.36	180098	51828	13024	58266	123119	-35002	21977	310569	128270
1991	13.99	10.09	193605	55870	14587	56703	127161	-40516	25928	336497	137735
1992	15.03	10.53	208126	58307	16338	54953	129597	-47433	31095	367592	149819
1993	16.16	10.99	223735	60854	18298	52992	132144	-54945	36646	404239	162881
1994	17.37	11.43	240515	63290	20494	50796	134581	-63214	42720	446959	177225
1995	18.68	11.88	258554	65782	22954	48337	137072	-72218	49264	496223	192772
1996	20.08	12.43	277945	68827	25708	45582	140118	-81768	56060	552282	209118
1997	21.58	13.07	298791	72371	28793	42497	143662	-91961	63168	615451	226420
1998	23.20	13.80	321201	76413	32248	39042	147704	-102872	70624	686075	244787
1999	24.94	14.54	345291	80511	36118	35172	151801	-114804	78686	764760	264780
2000	26.81	15.44	371187	85494	40452	30838	156785	-127427	86975	851736	285693
2001	28.83	16.25	399026	89980	45306	25984	161270	-141531	96225	947961	309047
2002	30.99	17.32	428953	95904	50743	20547	167195	-156251	105508	1053469	333049
2003	33.31	18.25	461125	101054	56832	14458	172344	-172806	115974	1169443	360071
2004	35.81	19.06	495709	105539	63652	7638	176829	-191266	127614	1297057	390170

PAYBACK (NO FINANCING) IN YEAR: 1990

PAYBACK (WITH FINANCING) IN YEAR: 1994

.

### FORCED AIR CONSUMER RETROFIT PAYBACK

COST COMPARISON FOR:FORCED AIR PRESENT YEAR: 1984 HOOKUP YEAR: 1985 TOTAL CONVERSION COST (PRESENT YEAR):\$140,000 ESCALATION RATE: 6.50% TOTAL CONVERSION COST (HODKUP YEAR):\$149,100 PERCENT FINANCED: 100% CASH INVESTMENT: \$0 TERM OF LOAN (YEARS): 20 Cost: \$140/kWt INTEREST RATE: 12.001 ANNUAL PAYMENT: \$19,961 TAX RATE: 50% EXPENSING DEDUCTION: \$5,000 YEARS OF DEPRECIATION: 5 ECIATION? (1=YES,0=NO): 1 POTENTIAL END USE ENERGY SAVINGS FROM CONVERSION: 0X ESTIMATED CURRENT SYSTEM EFFICIENCY: 55% HEATING EQUIPMENT TYPE: OIL BOILER ----- CURRENT ANNUAL FUEL USE -----PRIMARY BACKUP BACKUP TOTAL FUEL TYPE: OIL

	CONSUMPTION (MILLION	BTU):	13843	13843
ESTIMATED DH	CONSUMPTION (MILLION	BTU1:		7614
CURRENT	FUEL RATE (\$/MILLION	BTU1:	8.43	
	FUEL ESCALATION	RATE:	7.50%	

	CURRENT	DH		I	DISTRICT	HEATING -	I		DISTRICT		ENERGY
	FUEL	FUEL	CURRENT		CONVE	RSION	2	TAX	HEATING	CUMULATIVE	COST
	RATE	RATE	ENERGY	ENERGY	AMORTI	ZATION	TOTAL	EFFECTS	SAVINGS	SAVINGS	SAVINGS
YEAR	\$/MMBTU	\$/MMBTU	COSTS	COSTS	PRINC.	INTER.	COSTS	(\$)	(\$)	(\$)	(\$)
1985	9.06	12.33	125449	93876	2069	17892	113838	6467	18078	18078	31572
1986	9.74	12.98	134857	98825	2318	17644	118787	6657	22728	40806	36032
1987	10.47	8.08	144972	61518	2596	17366	81480	-17913	45579	86385	83453
1988	11.26	8.54	155845	65021	2907	17054	84982	-21754	49108	135493	90824
1989	12,10	8.74	167533	66543	3256	16705	86505	-27012	54017	189509	100990
1990	13.01	9.36	180098	71264	3647	16314	91225	-46260	42613	232122	108834
1991	13.99	10.09	193605	76822	4084	15877	96783	-50453	46369	278491	116783
1992	15.03	10.53	208126	80172	4575	15387	100133	-56284	51709	330200	127954
1993	16.16	10.99	223735	83674	5124	14838	103635	-62612	57488	387688	140061
1994	17.37	11.43	240515	87024	5738	14223	106985	-69634	63896	451584	153491
1995	18.68	11.88	258554	90450	6427	13534	110411	-77285	70858	522442	168104
1996	20.08	12.43	277945	94638	7198	12763	114599	-85272	78074	600516	183308
1997	21.58	13.07	298791	99510	8062	11899	119472	-93691	85629	686144	199281
1998	23.20	13.80	321201	105068	9029	10932	125030	-102600	93571	779715	216132
1999	24.94	14.54	345291	110702	10113	9848	130664	-112370	102257	881972	234588
2000	26.81	15.44	371187	117555	11327	8635	137516	-122499	111172	993144	253633
2001	28.83	16.25	399026	123722	12686	7276	143683	-134015	121329	1114473	275305
2002	30.99	17.32	428953	131868	14208	5753	151830	-145666	131458	1245931	297085
2003	33.31	18.25	461125	138949	15913	4048	158910	-159064	143151	1389082	322176
2004	35.81	19.06	495709	145116	17823	2139	165077	-174227	156405	1545486	350593

PAYBACK (NO FINANCING) IN YEAR: 1988

PAYEACK (WITH FINANCING) IN YEAR: 1989

#### STEAM AIR HANDLER CONSUMER RETROFIT PAYBACK

COST COMPARISON FOR: STEAM AIR HANDLERS PRESENT YEAR: 1984 HODKUP YEAR: 1985 TOTAL CONVERSION COST (PRESENT YEAR):\$250,000 ESCALATION RATE: 6.50% TOTAL CONVERSION COST (HOOKUP YEAR):\$266,250 PERCENT FINANCED: 100% CASH INVESTMENT: \$0 TERM OF LOAN (YEARS): 20 INTEREST RATE: 12.00% Cost: \$250/kWt ANNUAL PAYMENT: \$35,645 TAX RATE: 50% EXPENSING DEDUCTION: \$5,000 YEARS OF DEPRECIATION: 5 ACCELERATED DEPRECIATION? (1=YES,0=NO): 1 POTENTIAL END USE ENERGY SAVINGS FROM CONVERSION: 13% 55% ESTIMATED CURRENT SYSTEM EFFICIENCY: HEATING EQUIPMENT TYPE: OIL BOILER ----- CURRENT ANNUAL FUEL USE ------PRIMARY BACKUP BACKUP TOTAL FUEL TITEL OIL CONSUMPTION (MILLION BTU): 13843 13843 IMATED DH CONSUMPTION (MILLION BTU): 5814 FUEL TYPE: OIL 13843 ESTIMATED DH CONSUMPTION (MILLION BTU): CURRENT FUEL RATE (\$/MILLION BTU): 8.43 FUEL ESCALATION RATE: 7.50%

	CURRENT	DH		[	DISTRICT	HEATING	I		DISTRICT		ENERGY
	FUEL	FUEL	CURRENT		CONVERSION			TAX	HEATING	CUMULATIVE	COST
	RATE	RATE	ENERGY	ENERGY			TOTAL	EFFECTS	SAVINGS	SAVINGS	SAVINGS
YEAR	\$/MMBTU	\$/MMBTU	COSTS	COSTS	PRINC.	INTER.	COSTS	(\$)	(\$)	(\$)	(\$)
1985	9.06	12.33	125449	71687	3695	31950	107333	11188	29304	29304	53761
1986	9.74	12.98	134857	75466	4139	31507	111112	14795	38541	67845	59391
1987	10.47	8.08	144972	46978	4635	31010	82623	-6061	56288	124133	97994
1988	11.26	8.54	155845	49652	5192	30454	85297	-10438	60109	184242	106192
1989	12.10	8.74	167533	50815	5815	29831	86450	-16012	65060	249303	116718
1990	13.01	9.36	180098	54420	6512	29133	90065	-48273	41760	291063	125678
1991	13.99	10.09	193605	58664	7294	28352	94309	-53295	46001	337064	134941
1992	15.03	10.53	208126	61222	8169	27476	96867	-59714	51545	388609	146904
1993	16.16	10.99	223735	63897	9149	26496	99542	-66671	57522	446131	159839
1994	17.37	11.43	240515	66455	10247	25398	102100	-74331	64084	510215	174060
1995	18.68	11.88	258554	69071	11477	24168	104716	-82657	71180	581395	189483
1996	20.08	12.43	277945	72269	12854	22791	107914	-91443	78589	659984	205677
1997	21.58	13.07	298791	75990	14397	21249	111635	-100776	86380	746364	222801
1998	23.20	13.80	321201	80234	16124	19521	115879	-110723	94599	840963	240967
1999	24.94	14.54	345291	84536	18059	17586	120182	-121584	103525	944488	260754
2000	26.81	15.44	371187	89769	20226	15419	125414	-133000	112774	1057261	281418
2001	28.83	16.25	399026	94478	22653	12992	130124	-145778	123125	1180385	304548
2002	30.99	17.32	428953	100700	25372	10274	136345	-158990	133619	1314004	328254
2003	33.31	18.25	461125	106107	28416	7229	141752	-173895	145478	1459483	355018
2004	35.81	19.06	495709	110816	31826	3819	146461	-190537	158711	1618194	384893

PAYBACK (NO FINANCING) IN YEAR: 1989

PAYBACK (WITH FINANCING) IN YEAR: 1989